ORIGINAL PAPER



Is longer maternal care always beneficial? The impact of a 4-year paid parental leave

Alena Bičáková¹ · Klára Kalíšková^{1,2}

Received: 5 September 2022 / Accepted: 5 February 2024 / Published online: 11 March 2024 © The Author(s) 2024

Abstract

We study the impact of an extension of paid family leave in the Czech Republic from 3 to 4 years on children's long-term outcomes. We find that an additional year of maternal care at age 3 has an adverse effect on children's human capital investments and labor market attachment. Affected children are 6 percentage points less likely to be enrolled in college and 4 percentage points more likely to be not in education, employment, or training (NEET) at age 21–22. While the negative impact on education is persistent, with an 8 percentage points lower probability of completing college by the age of 27, the effect on NEET is short-lived. The results are driven by children of low-educated mothers, whose education and NEET outcomes are affected by as much as 12 percentage points. Our findings are consistent with previously documented positive effects of universal childcare on child long-term outcomes and with the fact that the extended maternal care induced by the extension of family leave led to a postponement of public kindergarten enrollment.

Keywords Family leave \cdot Maternal care \cdot Subsidized childcare \cdot Child outcomes \cdot Human capital \cdot Labor market attachment

JEL Classification $J13 \cdot J18 \cdot J21 \cdot J24$

Responsible editor: Terra McKinnish

Alena Bičáková alena.bicakova@cerge-ei.cz

> Klára Kalíšková klara.kaliskova@cerge-ei.cz

- ¹ CERGE-EI, a joint workplace of Charles University and the Economics Institute of the Czech Academy of Sciences, Politických vězňů 7, 111 21 Prague, Czech Republic
- ² Faculty of Informatics and Statistics, Prague University of Economics and Business, W. Churchill Sq. 1938/4, 130 67 Prague 3, Czech Republic

1 Introduction

Early childhood inputs have an important impact on children's development, human capital accumulation, and labor market performance later in life (García et al. 2020). While there is substantial evidence that early maternal care is the most beneficial for child outcomes (Rossin-Slater 2018), other forms of childcare also become important as children grow older. Preschool education has been consistently found to improve child outcomes, particularly for children from disadvantaged families (Felfe and Lalive 2018). Extended parental care that crowds out institutional childcare can have a negative impact on child development (Canaan 2022). The impact is likely to be greater for children of low-educated parents, who on average provide lower-quality childcare and tend to stay at home with children for longer periods (Pronzato 2009; Cohn et al. 2014). Policies offering well-intentioned family leave of several years may then impede intergenerational mobility and increase educational inequality.¹ Understanding the impact of prolonged parental care on long-term child outcomes is therefore crucial not only for the design of family leave policies but also for policies affecting socio-economic mobility and inequality. However, research on the effects of parental care at preschool age is scarce, as are the potential sources of exogenous variation in the duration of prolonged parental care.

This paper provides the first evidence on the impact of full-time maternal care beyond the child's 3rd birthday on children's human capital investment and labor market attachment later in life. We do so by exploiting the variation in the duration of maternal care induced by the extension of paid family leave from 3 to 4 years in the Czech Republic in 1995. We find that the extended maternal care had a substantial negative impact on children's long-term education and labor market outcomes, especially for children of mothers with lower levels of human capital.

The importance of full-time maternal care for child development decreases with the age of the child. While studies evaluating the impact of introducing or extending short family leave that increases exposure to maternal care during the child's first year of life find a positive or no impact on child outcomes (see, for example, Carneiro et al. 2015; Dustmann and Schönberg 2012; Huebener et al. 2019; Rasmussen 2010; Dahl et al. 2016), research analyzing the extension of longer family leave does not always confirm the positive impact of maternal care. The limited evidence suggests that non-parental inputs become increasingly important beyond the child's 2nd birthday, but knowledge about the effect of full-day parental care at later ages is lacking (see Rossin-Slater 2018; Huebener 2016 or Table A5 in Danzer et al. 2022 for an overview). While there is evidence of adverse effects of maternal care at the age of 2 on children's test scores and schooling choices (Canaan 2022; Dustmann and Schönberg 2012), there is no research on whether these effects persist into adulthood. To the best of our knowledge, our study provides the first evidence on the impact of paid family leave longer than 2 years on long-term child outcomes.²

¹ By family leave we mean maternity leave, parental leave, or a similar policy instrument designed to encourage parents to stay at home to care for their child.

 $^{^2}$ Danzer et al. (2022) is the only study other than ours to provide evidence on the impact of paid leave on long-term education and labor market outcomes, but the extension of leave in Austria they evaluate is only up to 2 years.

The 1995 reform increased the duration of paid parental leave from 3 to 4 years while leaving the amount of the flat-rate monthly allowance essentially unchanged.³ The take-up of paid leave effectively required that the parent did not work and that the child did not attend an institutional childcare facility. The extended leave option came into effect on October 1, 1995, and all parents with a youngest child aged 3 or less were eligible. We compare the outcomes of the first cohort of children affected by the 1995 reform with the last-unaffected cohort in order to estimate the impact of maternal care at the age of 3 induced by the family leave extension. We control for potential confounding factors with cohorts of older children, using the standard difference-in-differences methodology.

Our results suggest that the affected children are 6 percentage points (p.p.) less likely to be enrolled in college and 4 p.p. more likely to be not in education, employment, or training (NEET) at the age of 21–22. While the negative impact on education is persistent, with an 8 p.p. reduction in the probability of graduating from college by the age of 27, the effect on NEET is only short-lived. The effect is driven by children of low-educated mothers, whose education and NEET outcomes are affected by as much as 12 p.p. We find no effect of the reform on nest-leaving, family formation, or fertility at the age of 21–22. Our findings are robust to comparing the long-term outcomes of children in our treatment and control groups at the same age or at the same calendar time, and to using older cohorts as alternative control groups. Two placebo tests also confirm the validity of our findings.

Similar to the previous research (Dustmann and Schönberg 2012 and Canaan 2022), we do not observe the actual take-up of the leave by the mothers of the children whose outcomes we analyze, and therefore estimate the intention to treat (ITT) effect of the 1995 extension of family leave.⁴ We provide evidence on the size of take-up by estimating the impact of the reform on eligible mothers of the same cohorts of children whose long-term outcomes we analyze when observed at the age of 3.⁵ Eligibility was universal among mothers with a youngest child aged 3, and the size of take-up was larger than documented in most previous research, suggesting that the reform induced about 27% of mothers with a youngest child aged 3 to extend family leave beyond the child's 3rd birthday.⁶ We find no evidence of selectivity in take-up based on observable characteristics, including mothers' education, or pre-reform labor market conditions.

While 4-year family leave is rare, full-time maternal care beyond the child's 3rd birthday is not an exceptional phenomenon, especially among families with lower socio-economic status (Cohn et al. 2014). Understanding its impact on children is important for addressing issues of inequality and social mobility.⁷ However, mothers

³ The parental allowance is a non-means-tested benefit that starts either at the end of the maternity leave or immediately after childbirth if the mother is not eligible for maternity benefit.

⁴ While Danzer and Lavy (2018) also produce only ITT estimates, Danzer et al. (2022) observe the actual take-up of family leave and estimate the LATE using an IV approach. Their supplementary analysis of education outcomes on PISA scores is, again, restricted to the ITT estimates only.

⁵ Note that these are the same cohorts but different individuals, as we have repeated cross-sectional data.

⁶ Take-up of the treatment in the studies evaluating family leave extensions is typically much lower (e.g., less than 10% in Dustmann and Schönberg 2012).

⁷ Note that while the Czech family leave system has undergone many changes since 1995, the 4-year paid leave is still an option.

who stay at home for 4 years are typically self-selected and much more likely to either extend their leave further or withdraw permanently from the labor force (Bičáková 2016). This makes it impossible to assess the impact of full-time maternal care at age 3 on long-term child outcomes separately from exposure to maternal care at later ages. The institutional context of the Czech family leave reform of 1995 seems particularly useful for this research question. The Czech Republic is a country that combines very long family leave (with a very high take-up rate) and a traditionally strong overall attachment of women to the labor force (a legacy of the communist regime). While the 1995 reform induced a high share of mothers to stay at home with their children for more than 3 years, the majority returned to the labor force by the time their youngest child was seven (Bičáková and Kalíšková 2019). This allows us to interpret our findings as the impact of extending temporary (rather than permanent) maternal care beyond a child's 3rd birthday.

Prior to the extension of family leave in 1995, the majority of 3-year-old children were enrolled in public kindergartens. The reform thus postponed kindergarten enrollment by at least 1 year for most children whose mothers extended their parental leave. This relates our paper to a strand of literature that evaluates the impact of preschool education on child outcomes using universal childcare reforms (see Felfe and Lalive 2018 and Dietrichson et al. 2020 for an overview). Unlike existing family leave studies, which focus on children aged 0-2, this research also provides evidence on the impact of the type of care a child is exposed to between the ages of 3 and 6 on long-term child outcomes. It unanimously concludes that universal childcare has a positive effect on school progression, educational attainment and labor market outcomes, particularly for children from low socio-economic backgrounds (Dietrichson et al. 2020).⁸ Havnes and Mogstad (2011) examine the impact of a large-scale expansion of subsidized formal childcare primarily replacing (non-maternal) informal care for 3-6-year-olds on children's education and labor market outcomes. Our estimates of the impact of maternal care at age 3 that replaces care in public kindergartens turn out to be remarkably similar to that implied by their findings, in both sign and magnitude.⁹ The positive effect of attending formal childcare at age 3, driven mainly by children with loweducated parents, is also confirmed by Felfe et al. (2015) and Havnes and Mogstad (2015). Finally, Bailey et al. (2021) document a considerable positive effect of Headstart, a means-tested public preschool program for children aged 3-5 in the US, on college enrollment of the affected children and their likelihood of working as adults.¹⁰

⁸ Rude (2023) is one of the few exceptions finding that the positive impact of the comprehensive universal early childhood development program in Chile, with interventions from birth to age 5, on school education outcomes is smaller for the socio-economically vulnerable children.

⁹ Note that they study a reverse mechanism compared to ours, with formal care replacing informal.

¹⁰ Note that the take-up rate in the papers evaluating universal childcare reforms or similar interventions is somewhat higher (18 p.p. in Havnes and Mogstad 2011 or 15 p.p. in Bailey et al. 2021) but still below that documented here for the 1995 family leave extension. Moreover, preschool programs, such as Headstart, tend to be means-tested and focus on a subpopulation of children from lower socio-economic backgrounds. The universal eligibility and evidence of take-up spread across all demographic and socio-economic groups in our setting make our findings much more general.

While we attribute an important part of the negative impact of extended maternal care on the human capital accumulation and labor market outcomes of children to the postponement of enrollment in formal preschool, there may be other mechanisms at work behind our findings. In particular, we consider the indirect effects through maternal and other family outcomes, which are also likely to have been affected by the family leave reform and, at the same time, have been documented to affect child development (Canaan 2022). There is evidence of a negative, but only temporary, impact of the 1995 reform on mothers' post-leave labor market attachment, which disappeared by the time the children entered school (Bičáková and Kalíšková 2019). We explore the potential impact of the reform on long-term household income and stability by looking at mothers' labor market outcomes and marital status when the affected children were aged 11–17. While we find no evidence for these alternative channels, we acknowledge that the loss of 1 year of the mother's earnings must have had a substantial impact on household income and could also have affected the quality and quantity of inputs the child received at home during the extended leave.

Our paper contributes to three distinct strands of the literature: First, we extend the research assessing the impact of family leave reforms on child outcomes (as surveyed in Rossin-Slater 2018; Huebener 2016 or Table A5 in Danzer et al. 2022) by providing evidence on the impact of a 4-year-long paid family leave on long-term child outcomes. Second, we complement the studies of universal childcare reforms (as reviewed in Felfe and Lalive 2018) by estimating the impact of postponing formal preschool education until the age of 4. Thanks to the specific institutional setting, we are able to estimate the effect of extended maternal care against a well-defined and relatively homogeneous counterfactual.¹¹ Namely, we compare the impact of the two dominant forms of childcare at the age of 3 (maternal care versus preschool education) prior to and after the reform, thus providing new evidence connecting the two different strands of research. Finally, the documented impact of being exposed to an extra year of maternal care instead of attending public kindergarten at the age of 3 also contributes to research on the impact of early childhood inputs on long-term child outcomes along the lines of, for example, García et al. (2020). Although we use a fundamentally different methodology, our estimates can serve as inputs to the structural models of skill production used to analyze children's outcomes over their life cycle in this line of research.¹²

The remainder of the paper is organized as follows: Section 2 provides the institutional background, Section 3 describes the data and methodology, and Section 4 presents our estimation results. Section 5 discusses the identification assumptions and examines the robustness of our estimates. Section 6 considers potential mechanisms behind our findings, and Section 7 concludes.

¹¹ Public kindergartens, which were affordable and available in all municipalities with more than 2000 inhabitants and in 75% of smaller municipalities (Kuchařová et al. 2009), were attended by the majority of 3-year-olds before the reform. They were under the jurisdiction of the Ministry of Education and shared the same centrally determined curriculum. See Section 2.2 for further details.

¹² The different strands of the related literature are discussed in detail in Section 3 of Bičáková and Kalíšková (2022b), the working paper version of this study.

2 Institutional background

2.1 Family leave policies and the 1995 reform

Family leave policies in the Czech Republic include job protection, maternity benefits, and parental allowance. Parents are eligible for job-protected leave until their child's 3rd birthday.¹³ The job protection period was introduced in 1990 and has been maintained at 3 years duration since then. Mothers who have been employed for at least 270 days in the 2 years prior to a child's birth are entitled to maternity benefits for 28 weeks (starting 6 to 8 weeks prior to birth). The maternity benefit is 70% of the woman's salary from the last 12 months prior to the commencement of the maternity leave.

A parent caring for a child is also eligible for parental allowance, a non-meanstested flat-rate benefit. The parental allowance starts either immediately after maternity benefits end or immediately after childbirth if the mother is not eligible for maternity benefits.¹⁴ Parental allowance is only paid for the youngest child. When a mother gives birth to another child, any ongoing payments of allowance for the older child are automatically terminated.¹⁵

The eligibility criteria for the parental allowance at the time of the analysis required that the mother's earnings were below a certain threshold (in 1995, the threshold was CZK 1800 per month — less than one-fifth of the average female wage then) and that the child who entitled the parent to the allowance did not attend a childcare facility.¹⁶ Given these conditions and the very limited availability of part-time jobs in the Czech Republic, the receipt of a parental allowance in the 1990s implied that the parent was caring for a child at home and that the child was not enrolled in a childcare facility.

Until 1995, the receipt of parental allowance coincided with the 3-year period of job protection. In October 1995, the receipt of monthly parental allowance was extended until the child's 4th birthday, thus exceeding the unchanged duration of job protection by 1 year. All parents with children under the age of 3 on October 1, 1995, were eligible for the extended parental allowance. The extension of the parental allowance was unexpected and likely led to delayed rather than immediate adjustment by mothers.¹⁷ It was not announced until the end of May, by which time the first cohort of eligible mothers (those with children turning 3 in the fall after October 1, 1995) were likely to have already determined their return to work and their child's preschool enrollment for the next school year starting in September.

¹³ Employees with permanent contracts are entitled to job protection. An employee with a fixed-term contract is eligible for job protection up to the date of contract expiration.

¹⁴ There was no paternity leave until 2018, when a 7-day paternity leave was introduced. Parental allowance can, in principle, be received by a father but this is still very rare (0.78% of parental allowance recipients were fathers in 2001; earlier data are not available).

¹⁵ The remaining part of the total parental allowance for the older child is lost and the family receives only the allowance for the newborn. This rule has recently been changed (in 2021) and the remaining part of the allowance for the older child is now paid out as a lump sum after the birth of another child.

¹⁶ No conditions are imposed on childcare attendance of older siblings.

¹⁷ It was added to the Act on State Social Support during the legislative process. The bill was passed by Parliament on May 26, 1995.

There was also a slight increase in the monthly allowance payment from CZK 1740 to CZK 1848 in 1995, as part of the gradual valorization of the monthly allowance in the 1990s and early 2000s, but this change was negligible compared to the amount that parents gained by staying at home with a child for one more year and collecting an additional 12 months of parental allowance (see Fig. 1).¹⁸ The total allowance available to a mother of a newborn thus increased by CZK 26,064 (about 700 EUR) and for a mother of a child who just turned 3 by CZK 22,176 (about 600 EUR). Regarding the net income effect on a mother of a 3-year-old child, who would have returned to work after the 3-year leave prior to the reform and who earned an average wage: the extension of leave until the child's 4th birthday caused a net income loss of CZK 88,704 (about 2400 EUR), which corresponds to an 80% drop in annual income in the fourth year after childbirth.

With job protection ending at the child's 3rd birthday and the low level of parental allowance benefits compared to the loss of income due to the absence of mother's earnings, there were few financial incentives for Czech women to take the extended leave. Moreover, as discussed in Section 2.2, there were public kindergartens that provided affordable universal childcare of reasonable quality for children over the age of 3. However, there were other reasons that may have affected the extent of take-up, and the evidence presented below shows that the inactivity of mothers with the youngest child aged 3 increased, indeed, substantially after 1995.

First, there was a strong political and media campaign supporting the conservative norm of a mother as the primary caregiver and emphasizing maternal care as the most beneficial for preschool children that accompanied the 1995 reform.¹⁹ The emphasis on the traditional role of the mother in the family and the importance of maternal care may also have emerged as a reaction to the norms imposed by the communist regime, which required everyone to work (unemployment was artificially kept at zero), including mothers with small children that were enrolled in large state-run nurseries and kindergartens with high child-to-teacher ratios. The take-up of extended leave after 1995 could be a result of the pressure of the campaign, or it could reflect mothers' own pro-family preferences, which were suppressed under the communist regime or arose in response to it.²⁰

Second, job-protected family leave lasting several years may induce firms (especially in institutional settings where enforcement is not strong enough) to reduce the costs of long-term commitments by adjusting their organizational structure and offering women returning from leave less attractive or less stable jobs than they had before having children (see Schönberg and Ludsteck 2014, for example, for evidence on the higher probability of job loss among women who returned to work after job-protected

 $^{^{18}}$ The monthly allowance of CZK 1848 — about 50 euros — was equivalent to about one-fifth of an average wage in the economy.

¹⁹ As religion was persecuted under communism, a return to Christian values, including conservative gender roles in the household, was a natural response in part of society after the collapse of the regime.

²⁰ Note that the decline in female labor force participation (which exceeded the EU-15 average during communism) was observed in many post-communist countries in the first decades after the end of the communist regimes (Bičáková and Kalíšková 2022a).



Fig. 1 Parental allowance in the Czech Republic. Note: The left vertical axis shows the total amount of parental allowance per one child (in CZK). The right axis shows the monthly parental allowance amount as a share of the average wage in the economy in a given year

leave).²¹ Moreover, the job protection may have been also less effective in transition economies because of the unstable business environment, in which firms disappeared quickly or jobs were rationed (Kantorova 2004; Fodor 2005).

Finally, the transition period was accompanied by important structural changes in the labor market, which may also have increased the incentive for mothers with less favorable job prospects to take up the extended leave. However, the unemployment rate in the Czech Republic was still very low in 1995, starting from zero at the end of communism. We formally test this hypothesis of selective take-up driven by local labor market conditions in Section 4.1 but find no empirical support for it.

Whatever the dominant reason, the increase in inactivity among mothers with a 3-year-old youngest child was indeed substantial after the 1995, when the option of an extended leave came into force. Figure 2, which depicts the evolution of the share of inactive mothers by the age of their youngest child, suggests that an additional almost 40% of mothers whose youngest child was 3 stayed at home until the child's 4th birthday in 1996 compared to 1994. This is only descriptive evidence, however, and may mask other underlying trends. We formally document the size of the take-up of the additional year of parental allowance in Section 4.1, where we estimate the impact of the 1995 reform on the inactivity of mothers with a youngest child aged 3, as a "first-stage" of our ITT analysis of long-term child outcomes.

Figure 2 also documents a slight increase in the inactivity of mothers with 2-year-old children after 1995. This rise could also be due to the reform. Prior to the introduction of extended leave, mothers of 2-year-olds who turned 3 by the end of the calendar year could enroll their children in kindergarten before their 3rd birthday, at the beginning of the school year in September, if there was sufficient capacity (for details on types and use of childcare, see Section 2.2) and return to work.

 $^{^{21}}$ Schönberg and Ludsteck (2014) find that 7% of women who return to work at the end of the job-protected leave become unemployed 1 or 2 months later.



Fig. 2 Share of inactive mothers by the age of their youngest child, 1993–1998.Note: The figure illustrates the share of mothers that are economically inactive (out of labor force) by the age of their youngest child in the 1990s. Source: Czech Labor Force Survey data (1993–1998), own calculations weighted by population weights

Finally, there was also a slight increase in inactivity among mothers of 4 and 5year-old children, suggesting that some mothers extended their leave even beyond the 4 years of paid parental leave.²²

As a final piece of descriptive evidence on the use of extended leave, Fig. 3 addresses the question of whether all mothers who complied with the 1995 reform and extended their leave beyond the child's third birthday stayed at home for a similar length of time or whether the increase in the share of inactive mothers of 3-year-olds masks heterogeneity in leave extension. The shares of inactive mothers by the age of their youngest child, measured in quarters of a year (the most detailed age information we can derive from the data) in 1994 and 1996 move almost in parallel between the ages of 3 and 4, suggesting that all mothers who extended their leave in response to the 1995 reform stayed at home for about three quarters of their child's fourth year or more.

2.2 Childcare types and usage

Public childcare in the Czech Republic consists of nurseries for children aged 1–2 and kindergartens for children aged 3–5. The number of nurseries, which operated as medical facilities under the jurisdiction of the Ministry of Health, was substantially

²² For the impact of the 1995 reform on the labor market outcomes of mothers with children aged 3–5 see also our earlier work (Bičáková and Kalíšková 2019).



Fig. 3 Share of inactive mothers by the age of their youngest child in quarters. Note: The figure illustrates the share of mothers that are economically inactive (out of the labor force) by the age of their youngest child (reported in quarters of a year) in the 1994 (before the reform) and 1996 (after the reform). Source: Czech Labor Force Survey data (1994, 1996), own calculations weighted by population weights

reduced in the early 1990s.²³ Nursery closures were accompanied by a wider shift in family policy priorities in the 1990s, discussed above, which encouraged women to stay at home after childbirth to raise their children (Saxonberg and Sirovátka 2006).

Public kindergartens are part of the education system, subject to similar rules as other schools, under the jurisdiction of the Ministry of Education, Youth and Sports. The preschool education provided by the public kindergartens must follow specific guidelines (curriculum) set by the Ministry of Education, Youth and Sports. The aim is to ensure that the child masters the basics of key competencies (learning, problem solving, communication, social and personal skills, etc.) at an early age and thus acquires the prerequisites for lifelong learning, in order to be more successful in the knowledge society. Children in kindergartens are organized in classes, which are usually made up of children of the same age. They typically operate on weekdays (with a 4–8 week break in July and August) for a maximum of 10h (7 am to 5 pm) per day.²⁴ The majority of public kindergartens charge a monthly fee.²⁵ Since public kindergartens are largely subsidized by the government, the cost is extremely low.²⁶ The maximum class size is 24 children. Preschool education in kindergartens

 $^{^{23}}$ While 15% of children aged 1–2 attended nurseries in 1990, the figure was only 5% in 1992, and only 1% by the end of the 1990s.

²⁴ See https://www.msmt.cz/areas-of-work/preschool-education?lang=2 for more detailed description of public kindergartens in the Czech Republic.

²⁵ Some of the kindergartens in less populated regions may be free of charge, by the decision of the director.

²⁶ In 2006 (earlier data are not available), the kindergarten fees were at most 1000 CZK per month including food and represented only 2.7% of the net income of an average family (Kuchařová et al. 2009).

is provided by teachers who have completed secondary or higher education with a diploma specializing in preschool pedagogy. There are 2 teachers per class, resulting in a child-to-teacher ratio of less than 12.²⁷

Unlike nurseries, the number of kindergartens was much less affected by the conservative policies of the early 1990s. They were both financially affordable and widely available to the vast majority of parents.²⁸ The slight reduction in the supply of kindergartens in the 1990s was more than offset by a sharp drop in fertility, resulting in a steady increase in overall kindergarten coverage for children aged 3 to 5 during the 1990s (see the gray bars in Fig. 4).²⁹

As the eligibility for parental allowance benefits required that the child was not enrolled in a childcare facility, the duration of paid parental leave and its use had a direct impact on the attendance of the youngest child at nurseries and kindergartens. Figure 4 shows the evolution of the total number of children attending public kindergartens by age on September 1 of a given year. Before the 1995 reform, the majority of 3, 4, and 5-year-old children attended kindergarten. While kindergartens are designed primarily for 3–5-year-olds, children between the ages of 2 and 6 may attend.³⁰

After the 1995 reform, children of mothers who took the extended family leave for the 4 years could start kindergarten only at the age of 4. The sharp drop in kindergarten enrollment by over 25% for 3-year-olds and by almost 50% for 2-year-olds, between 1994 and 1996 in Fig. 4 shows that the 1995 reform substantially affected not only mothers' inactivity but also their children's kindergarten enrollment.³¹

Note that the increase in mothers' inactivity does not necessarily correspond to the change in kindergarten attendance and vice versa. Part of the observed increase in inactivity may be driven by mothers replacing other informal care for their youngest 3-year-old child with their own care by taking advantage of the extended leave after 1995, without affecting the share of children in kindergarten.

The observed increase in mothers' inactivity and the decrease in children's kindergarten enrollment implied by the descriptive figures presented above also differ due to the fact that the two measures are based on different datasets collected at different points in the year. Namely, while mothers' inactivity in Fig. 2 as well as the "first-stage" results in Section 4.1 are based on the sample of women with the youngest child aged

 $^{^{27}}$ In 1994, the average class size was about 23 and the average child-to-teacher ratio was about 12 (Kuchařová et al. 2009). Unfortunately, there are no other measures of the kindergartens' quality from the 1990s. Given the centralized guidelines they have to follow, we do not expect substantial differences in the type of care provided by different institutions.

 $^{^{28}}$ Kuchařová et al. (2009) documents that in 2002 (the earliest statistics available) kindergartens were available in all municipalities with more than 2000 inhabitants and in 75% of smaller municipalities.

²⁹ The 1990s were marked by a sharp decline in fertility rates in most transition economies (Sobotka 2003), the Czech Republic being no exception. Sobotka (2003) argues that in Central Europe, the decline in fertility was mainly caused by postponement of parenthood (the average age of a Czech woman at first birth rose from 22.6 in 1993 to 24.6 in 1999).

 $^{^{30}}$ As the school year begins in September, children who will turn 3 by the end of the calendar year may enroll in kindergarten at the beginning of the school year when they are still 2, provided there are enough places. Children who are not physically or mentally mature enough can postpone their school entry age (the conditions for such postponement are defined by law), so there can be 6+ year-olds attending kindergarten.

 $^{^{31}}$ Although the observed decline in kindergarten enrollment also partly reflects the gradual decline in fertility, the population of 2- and 3-year-olds declined by only 8% between 1994 and 1996.



Fig. 4 Kindergarten attendance by age. Note: The figure shows the total number of children enrolled in public kindergartens by their age (age is reported as of September 1 in each calendar year) on the left vertical axis. The right vertical axis illustrates the ratio of children aged 3–5 enrolled in public kindergartens to all children aged 3–5 in the population (shown in grey bars).

Source: Ministry of Education, Youth, and Sports and Czech Statistical Office, own calculation

3 (those directly affected by the 1995 reform in relation to that particular child, in order to better isolate the effect of the leave extension), the aggregate kindergarten participation rates in Fig. 4 refer to all 3-year-old children. Second, the information on mothers' inactivity and the data on kindergarten enrollment are measured at different times of the year: while the former is measured for mothers whose youngest child is 3 years old at the time of the survey, the kindergarten enrollment rates are calculated on the basis of child's age as recorded on September 1. Part of the increase in mothers' inactivity should therefore also be compared with the decrease in the enrollment of 2-year-olds (who turn 3 in this school year).

Interestingly, the total number of children aged 4 and 5 attending public kindergartens remained fairly stable over the period (Fig. 4), suggesting that the children who stayed at home with their mothers at ages 2 and 3 after the 1995 reform also attended kindergarten, but only at a later age.³² The share of children aged 6 and over in kindergarten did not change much either, suggesting that the postponement of kindergarten enrollment after the 1995 reform did not result in later school enrollment.

The evidence presented in Fig. 4 shows that the 1995 reform shifted enrollment in public kindergartens from age 3 to age 4 and shortened the total time spent in

³² For example, in the first cohort of affected children (born between October 1992 and September 1993), only 55% of children attended kindergarten at age 3, but this quickly rose to 82% and 91% at ages 4 and 5, respectively.

preschool education, rather than postponing the school entry age or excluding some children from preschool education altogether.

During the 1990s, the overall trend in the share of children enrolled in kindergartens among 3–5-year-olds remained fairly stable from 73 to 80% (Fig. 4), with the slight increase mostly reflecting the drop in the denominator due to the decline in long-term fertility. The consistently high shares confirm that preschool education in public kindergartens was the dominant form of childcare for this age group.

Alternative forms of childcare for children aged 3–5 were still very rare at the end of the 1990s. Kuchařová et al. (2009) surveyed a representative sample of Czech municipalities in 2002 and found that in the 497 municipalities, there were only 3 private kindergartens, 8 childcare agencies, and 4 self-help organizations focused on childcare.³³ Another survey of parents in 2005 confirms the lack of availability and use of private childcare — only 2% of parents with children aged 1–10 reported that they sometimes used a childminder to help with childcare and most of them used this form of childcare only sporadically, not on a daily basis (Ettlerová et al. 2006). It was more common to use the help of grandparents or other relatives - 20% of parents with children aged 1–10 reported that grandparents helped them with childcare on a regular basis during the week, but it is unlikely that they would provide full-time daily care (Ettlerová et al. 2006).³⁴

The limited information on the use of alternative forms of childcare in the 1990s suggests that the majority of children whose mothers decided to stay at home for 4 years instead of 3 after the 1995 reform would otherwise have attended public kindergarten at age 3. Therefore, care in public kindergartens represents the main counterfactual to the extended maternal care whose effects we estimate.

2.3 Education system overview

Compulsory education in the Czech Republic lasts 9 years, from age 6 to 15, and consists of primary and lower secondary education (Table 1). Almost everyone completes lower secondary school (about 99% in our sample). High school completion is also very high (over 90% in our sample). Depending on the type of school, students complete high school between the ages of 17 and 19. Students typically enroll in tertiary education at the age of 19, after graduating from Gymnasium, the general high school corresponding to the academic track. A much smaller proportion comes from technical schools.³⁵

Tertiary education consists of post-secondary vocational education (2–3 years) and standard higher education. The usual structure of higher education consists of bachelor's degree programs (3–4 years), master's degree programs (1–3 years), and doctoral programs (3–4 years). However, there are also so-called unstructured programs, such

 $^{^{33}}$ Earlier data are not available, but it is very likely that these organizations were even more scarce in the 1990s when the market was still emerging.

³⁴ Given the high share of children aged 3 enrolled in kindergartens before the 1995 reform, the grandparental care was probably used mostly for picking up children from kindergartens and for afternoon activities.

 $^{^{35}}$ There are also conservatoires which provide arts education that may span both secondary and tertiary education and last 6–8 years.

Age	Level of education	ISCED	Type of school
6–10	Primary	ISCED 1	Basic School
11–14	Lower Secondary	ISCED 2	Basic School or Gymnazium
15-18	Upper Secondary	ISCED 3	Vocational (2–3 years)
			Technical (4 years)
			Gymnazium (4 years)
19–27	Tertiary Education	ISCED 5, 6	Professional (2-3 years)
			Structured Higher Education
			(BA 3-4 years, MA 1-3 years)
			Unstructured Higher Education (4-6 years)
			Doctoral Programs (3-4 years)

Table 1 Overview of the Czech education system

Note: Gymnazium is a General Secondary School, starting at the age of 11, 13, or 15, and lasting 8, 6, or 4 years

as medicine, law, or some art schools, which last 4–6 years and do not award bachelor's degree. Tertiary education can therefore be completed at any time between the ages of 21 and 24, depending on the type of school. While students enrolled in professional tertiary programs or structured tertiary programs can obtain their first degree the age of 21, students enrolled in unstructured programs can complete their education at the age of 23 at the earliest.³⁶

Education is primarily provided by public institutions (including public universities at the tertiary level) and is free. Paid private schools exist at all levels of the education system but public schools serve over 90% of students.³⁷

3 Methodology

3.1 Data

For the main analysis of the impact of the 1995 reform on the education and labor market outcomes of the affected children in their early twenties, we use the Labor Force Survey (LFS) for the Czech Republic for the years 2010–2016.³⁸ Because the data do not contain information on the duration of maternal care the child was exposed to or the number of years the child attended preschool, we can only estimate the ITT effect of the 1995 reform. To explore the size of the treatment, we use the 1994–1997

³⁶ It is quite common for students to prolong their tertiary studies beyond the standard number of years, which then leads to further delays in completing tertiary studies.

³⁷ Ministry of Education, Health and Sports of the Czech Republic: https://dsia.msmt.cz//vystupy/ vu_vs_f1.html.

 $^{^{38}}$ When discussing the persistence of the estimated effect into their late twenties, we use the most recent LFS data for the years 2014–2020.

LFS data to estimate the impact of the 1995 reform on the probability that a mother is at home with a 3-year-old child (see Section 4.1).³⁹

We focus on child outcomes at age 21–22 for two reasons: First, we can already observe the long-term measures, including high school completion rates, the share of individuals in tertiary education, and the labor market outcomes of recent high school graduates who have not enrolled in college and of children with less than high school education. Second, the share of children still living with their parents at this age is sufficiently high, which is crucial for our heterogeneity analysis of the impact of the reform by mothers' education, since information on parental background is only available for them in the data. In the affected cohorts, about 75% of individuals reside in the same household as their mother (see Online Appendix Table A.1 for the exact share in the cohorts we use in the analysis). Although the share is relatively high, potential selection into nest-leaving may still bias our estimates from the heterogeneity analysis. We provide evidence of a zero impact of the reform on the probability of residing with one's mother at age 21–22 in Section 5.

The identification of the last cohort unaffected by the reform and the first cohort affected is based on the exact age of the child. Since the LFS data provide age information only in full years, we use a rotational panel structure of the data and derive age in quarters of a year based on the changes observed between the two consecutive quarters.⁴⁰

3.2 Estimation strategy

We estimate the ITT impact of the 1995 reform using a difference-in-differences method, as is common in the literature (see, e.g., Dustmann and Schönberg 2012; Baker and Milligan 2015; Carneiro et al. 2015; Dahl et al. 2016; Danzer and Lavy 2018). We compare the outcomes of two cohorts of children: the last-unaffected cohort and the first cohort affected by the reform. The 1992 cohort of children (children born before October 1, 1992) was the last-unaffected cohort because mothers of children who turned 3 before October 1995 had already completed their 3 years of paid parental leave before the reform. The 1993 cohort (children born on or after October 1, 1992) were the first children whose mothers were eligible for paid parental leave until the child's 4th birthday.⁴¹

We compare the outcomes of the unaffected and affected cohorts at the same age (in quarters) when they are 21–22 years old.⁴² Fixing age ensures that our results are

1, 1992, who were affected by the policy at age 3, were fixed well before the reform was announced.

³⁹ When discussing the mechanisms behind the estimated impact of the reform on long-term child outcomes, we also use the 1999–2011 LFS data to explore the potential impact of the reform on mothers' outcomes and family composition at an intermediate stage, when the affected children are aged 10–17.

⁴⁰ Each individual is observed for five consecutive quarters in the data, and an observed change in age helps to determine the quarter-of-birth. We drop less than 9% of children in the sample for whom a quarter-of-birth cannot be derived due to attrition, misreporting, or measurement error due to irregular interview timing.

⁴¹ Note that birth date manipulation is impossible in our setup, as we compare cohorts born around October 1, 1992, while the reform was announced in May 1995. Thus, the birth dates of children born after October

 $^{^{42}}$ We use the 2-year age window to ensure a sufficiently large sample size but control for age differences within this window with quarter-of-birth and calendar quarter-year fixed effects.

not driven by age effects, but also implies that the outcomes of the two cohorts are observed at different calendar times and are therefore affected by different business cycle conditions and seasonality effects. In addition, the last-unaffected and firstaffected cohorts may be born in different quarters of the year, so their outcomes may also differ due to quarter-of-birth effects.

Following the previous studies on the effects of family leave policies on child outcomes (Liu and Skans 2010; Rasmussen 2010; Dustmann and Schönberg 2012; Carneiro et al. 2015; Danzer and Lavy 2018), we control for any confounding factors, such as the quarter-of-birth or business cycle effects, using two consecutive cohorts from a neighboring year in which no policy change occurred as the counterfactual. Our identification strategy is illustrated in Fig. 5, where we clarify which cohorts are in the treatment and control groups, the before and after periods, and what we refer to as the last-unaffected and the first-affected cohort.

We define four different treatment groups based on the size of the window around the cut-off date of birth — Treat 1 are children born within one quarter before/after the October thresholds, while Treat 4 are those born within four quarters around the thresholds (see Online Appendix Table A.2 for the exact definitions of our treatment and control cohorts in the four specifications using Treat 1 to Treat 4 groups).

We present our baseline results for the four different samples, varying the width of the window around the cut-off, to explore how our estimates change with distance from the reform date and with the number of observations. The last-unaffected and first-affected cohorts in the narrower windows are closer in terms of birth and should therefore be more similar, which may help to better isolate the effect of the 1995 reform from other potential differences between cohorts that would not be captured by the control cohorts. However, the samples in the narrower windows are relatively small and some of the outcomes analyzed are quite rare (resulting in few individuals with lefthand-side variable equal to 1). Therefore, our preferred specification is the one with the broadest window. We note two other advantages of Treat 4 over the other samples: First, the last-unaffected and first-affected cohorts from 1 year before and after the reform contain birth dates evenly distributed across all four quarters of the year, so the quarter-of-birth effects should be similar in the two groups, whereas, for example, Treat 1 compares only third and fourth quarters and relies on the control groups to capture the quarter-of-birth effects. Second, the 1995 reform was likely to induce



Fig. 5 Identification strategy

delayed rather than immediate adjustment by mothers, as discussed in Section 2.1. The policy change was not announced until May 1995, by which time mothers with children turning 3 in the Fall of 1995 were likely to have already made decisions about returning to work and enrolling their child in preschool. The reform would have no impact on these mothers of children in the narrowest specification (Treat 1).

Using the standard difference-in-differences methodology, we estimate the following equation:

$$y_i = \alpha_0 + \alpha_1 Treat_i + \alpha_2 After_i + \alpha_3 Treat_i * After_i + X_i \beta + u_i$$
(1)

where y_i is the outcome variable. We use several outcome variables: completed high school,⁴³ tertiary (being a student in or having completed tertiary education),⁴⁴ NEET (not in education, employment, or training), employed, unemployed, housework (inactive with housework as main activity).

 $Treat_i$ is the dummy for treated cohorts: children born before/after October 1992 have $Treat_i = 1$, while children born before/after October 1990 have $Treat_i = 0$. $After_i$ is the dummy for being born after the October thresholds — children born after October 1990/1992 have $After_i = 1$ and children born before the October thresholds have $After_i = 0$. The coefficient of the interaction term (α_3) captures the effect of the 1995 reform.⁴⁵

The control variables (X_i) are a dummy variable for gender, fixed effects for a quarter-of-birth, fixed effects for the current region of residence (NUTS 4), and fixed effects for all calendar quarter-year combinations.⁴⁶ In addition to our baseline specification, we also explore the heterogeneity of the impact of the 1995 reform on child outcomes by mother's education and child's gender.

In our main specification, we compare the outcomes of the first-affected and lastunaffected cohorts (born around October 1, 1992) observed at the same age (21–22), controlling for potential quarter-of-birth and business cycle effects with the control cohorts (born around October 1, 1990) observed in the same calendar year quarters as the corresponding treatment cohorts.⁴⁷ Summary statistics for the treatment and control cohorts can be found in Online Appendix Table A.1.

 $^{^{43}}$ We do not consider completion of lower secondary education as an outcome because over 99% of our treatment and control groups have completed lower secondary school (see Online Appendix Table A.1).

⁴⁴ Most individuals in our treatment groups have not (yet) finished tertiary education (only about 6-7% do, see Online Appendix Table A.1), so we use a variable that captures either completion of tertiary education or current tertiary study (about 45% of treated individuals are studying in tertiary education) to proxy for the reform's impact on college enrollment. The overall share of individuals who have completed tertiary education or are currently studying is similar for our control cohorts, but they are more likely to have completed tertiary studies (Online Appendix Table A.1).

⁴⁵ We discuss the theoretical framework of our analysis, including the mechanisms and potential direct and indirect effects of the paid family leave extension, and formulate our hypotheses about the expected impacts of the 1995 reform on child long-term outcomes in Section 4 in Bičáková and Kalíšková (2022b), the working paper version of this study.

⁴⁶ Note that in our setup, including both calendar quarter-year fixed effects and age quarter-year fixed effects is not possible due to perfect multicollinearity. Depending on the specification, the calendar quarter-year fixed effects capture unobserved differences across calendar quarter-years and age.

⁴⁷ These cohorts include children of mothers who were all eligible for 3 years of paid parental leave and were not directly affected by the 1995 reform or any other earlier family policy change.

Our identification strategy is based on the assumption that the potential confounding factors are the same for the treatment and control cohorts, which are observed in the same year but at different ages. In particular, while outcomes are allowed to vary with economic conditions and/or quarter-of-birth effects, the impact of these confounding factors on outcomes must be age invariant.⁴⁸

While our main specification compares the outcomes of the first-affected and lastunaffected cohorts at the same age, we could also fix the calendar year for the firstaffected and last-unaffected cohorts and use the control cohorts observed at the same age to control for age-related confounding factors. To check the sensitivity of our results to our identification assumptions, we also estimate this alternative specification. See Section 5.1 for a detailed discussion of this second approach, its identification assumptions, and a comparison of the results with the main specification. To examine our identification assumptions and the validity of our findings, we document the trends in cohort outcomes before and after the reform in Sections 5.2 and 5.3 and conduct two placebo-type tests in Section 5.5.

4 Results

4.1 Impact on mothers

We do not observe the use of family leave by the mothers of the children whose longterm outcomes we analyze, so we can only estimate the ITT effect of the 1995 reform on children. We are, however, able to document the actual family leave take-up by a representative sample of mothers of the last-unaffected and first-affected cohorts of children from earlier surveys of the LFS data in 1994–1997, when these children were 3. The analysis of the impact of the 1995 reform on mothers' leave-taking can be considered as a first-stage of our estimation of the reform's ITT effect on long-term child outcomes.

To provide information on the size of the treatment, we estimate the impact of the 1995 reform on mothers' inactivity when their youngest child is 3. Specifically, we classify the mothers of children belonging to the treatment and control groups before and after the reform, as described in Online Appendix Table A.2, according to when their child was born.

Using the same empirical specification and DID strategy as described in Section 3.2 for estimating the impact on child outcomes, we estimate equation 1 on mothers, with the probability of being inactive as the explained variable and the same set of right-hand-side variables, just replacing child characteristics with those of mothers.⁴⁹

⁴⁸ To give an example, if the probability of employment of young individuals is affected differently by changes in economic conditions at age 21 than at age 23, the differences in outcomes of the last-unaffected and first-affected cohorts due to business cycle effects may not be correctly captured by our control group cohorts.

⁴⁹ In particular, the regressions control for mother's age, education, marital status, number of children, and presence of elderly in the household.

	Mother inac	tive when the youn	gest child is 3	
	Treat1	Treat2	Treat3	Treat4
Panel A: Baseline				
Treat*After	0.088	0.134***	0.261***	0.267***
	(0.067)	(0.050)	(0.043)	(0.039)
Observations	693	1201	1616	1970
Mean of dep. var	0.266	0.268	0.269	0.254
Panel B: By mothers' educe	ation			
Treat*After	0.114	0.212***	0.310***	0.299***
	(0.102)	(0.075)	(0.064)	(0.059)
Treat*After	-0.055	-0.151	-0.092	-0.059
*HighEducation	(0.133)	(0.099)	(0.085)	(0.079)
Observations	693	1201	1616	1970
Mean of dep. var	0.266	0.268	0.269	0.254
Panel C: By regional unem	ployment			
Treat*After	0.165*	0.211***	0.328***	0.311***
	(0.096)	(0.070)	(0.060)	(0.056)
Treat*After	-0.129	-0.140	-0.125	-0.080
*HighUnemployment	(0.132)	(0.099)	(0.085)	(0.079)
Observations	693	1201	1616	1970
Mean of dep. var	0.266	0.268	0.269	0.254

Table 2 Estimation results: impact on mothers

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1 (Panel A) and Treat*After interacted with a dummy variable for high-educated women (Panel B) and Treat*After interacted with a dummy for regions with the above-median unemployment rate in the pre-reform year of 1994 (Panel C). Results are presented for four different treatment and control groups (Treat1-Treat4), which differ by the size of the window around the cut-off birth date — October 1992 (see Online Appendix Table A.2 for details). All regressions control for mothers' age, education, marital status, number of children, presence of elderly in the household, and regional and quarter-year fixed effects. Standard errors are in parentheses. *p < 0.1; *p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (1994-1997), own calculations

The results in Table 2, which presents estimates of the impact of the 1995 reform on the inactivity of mothers when their youngest child is 3 for the four different windows (Treat 1 to Treat 4), reveal that mothers' compliance with the 1995 reform was indeed substantial. Depending on the specification, an additional up to 27 percent of mothers of the 3-year-olds extended their leave beyond their child's 3rd birthday in response to the 1995 reform.⁵⁰ The zero impact in the narrowest window and the fact that the size of the effect increases with the width of the window is likely to be driven by the

⁵⁰ Note that the impact of the reform documented in Bičáková and Kalíšková (2019) is even larger, suggesting a take-up of almost 40 p.p. among mothers of 3-year-olds. While here we estimate the impact on the very first cohort of children affected by the reform (born after October 1992), and show the take-up by their mothers, in our earlier work we focus on the mothers of the second cohort of affected children (born after October 1993), whose take-up was even larger, given the gradual adjustment of mothers to the reform, but whose children are less comparable to the last-unaffected cohort.

gradual adjustment of new cohorts of mothers whose children turned 3 after the 1995 reform was announced as discussed in Section 2.1.⁵¹

Our first-stage regression suggests that the extended leave was taken by more than a quarter of all mothers with a youngest child aged 3. However, there may be selection into the treatment, as these mothers are likely to differ in terms of, for example, work-life balance preferences from those who did not take the extended leave or from those who would have been at home with a youngest child aged 3 even in the absence of the reform.⁵²

To shed more light on selection into treatment, Table A.3 in the Online Appendix compares the mean characteristics of inactive mothers with a youngest 3-year-old child prior to and after the reform. In our data, there are twice as many inactive mothers of 3-year-olds after the reform as before. The differences in the summary statistics for the two groups are only minor and not statistically significant.

The only characteristic for which we observe a statistically significant difference is the number of siblings, implying that mothers who took the extended leave in response to the 1995 reform had, on average, fewer children than mothers who stayed at home with the youngest child aged 3 prior to the reform. This seems to be in line with our expectations that it was mainly mothers with strong pro-family preferences (as indicated by a higher number of children) who stayed at home with a 3-yearold before the reform. The 1995 reform may also have induced mothers with fewer children, including those with an only child, to stay at home for more than 3 years.

We next test selection into take-up more formally, using our first-stage regression of the impact of the 1995 reform on the inactivity of mothers with the youngest child aged 3. We examine potential heterogeneity in take-up by allowing the impact of the reform to vary by observable characteristics and test the statistical significance of the coefficients on the respective interaction terms.

We first examine two key factors that are likely to affect selection into take-up: mothers' education and local labor market conditions (see Table 2). More educated mothers are likely to earn more and therefore face a higher opportunity cost of not working for another year. They are, however, also more likely to have more educated partners with higher incomes and so can afford to stay at home for one more

⁵¹ This is also why the impact of the reform estimated in Table 2 differs from the change in mothers' inactivity of mothers implied by Figs. 2 and 3. First, the simple descriptive statistics presented in these figures are not directly comparable to the DID estimates in Table 2. Second, the DID estimates do not measure the overall increase in inactivity among mothers with the youngest 3-year-olds between 1995 and 1996, but only among mothers with the youngest 3-year-olds who turned 3 between Q4 1995 and Q3 1996. Given the gradual adjustment to the reform, the take-up of extended leave was even higher among mothers of children who turned 3 in Q4 1996 — the cohort captured by the descriptive Figs. 2 and 3, but not by our DID estimates.

⁵² Before the reform (in 1994), one quarter of mothers with a 3-year-old youngest child were inactive (see Fig. 2).

year. However, we find no heterogeneity in the extended leave take-up by mother's education. 53

Mothers who are not subject to job protection or whose previous jobs no longer exist may be more likely to extend leave (rather than become unemployed) when jobs are scarce. We test for selection into take-up based on mothers' job prospects by allowing take-up to vary by local labor market conditions prior to the reform, proxied by the regional unemployment rate in 1994. Interacting the reform effect with a dummy variable for regions with an unemployment rate above the median in 1994, we again find no support for selective take-up along this dimension.⁵⁴

In addition to the results on selectivity by mother's education and by local labor market conditions, we also estimate a series of similar regressions in which we interact the effect of the 1995 reform with other characteristics available in the data, one at a time, such as marital status, number of children, spouse's education, presence of elderly in the household, or the gender of the youngest child. None of the interaction terms, including the one with the number of siblings, turn out to be statistically significant (see Table A.4 in the Online Appendix).⁵⁵

The takeaway from our first-stage results for our ITT analysis of the impact of the 1995 reform on long-term child outcomes is therefore that the treatment was substantial (over 25 p.p. of mothers extended their leave from 3–4 years) and somewhat delayed, gradually increasing during the first year. Thus, the impact of the treatment should be most pronounced in our broader specifications (Treat 3 and Treat 4). Finally, we find no evidence of selectivity in take-up based on mothers' characteristics and pre-reform local labor market characteristics. The information in the dataset is, however, limited and there may be other (unobserved) aspects, such as pro-family preferences, in which mothers who complied differed from those who did not.

4.2 Education outcomes of children

The main estimation results for education outcomes are presented in Table 3. The outcomes are described by two binary variables — one for high school completion and the other for enrollment in or completion of tertiary education. Our baseline specification reveals a weak negative impact of the family leave reform on high school completion (for Treat 2 and 3 specifications, see Panel A of Table 3) and a stronger negative impact on tertiary studies in our broader specifications (Treat 3 and 4, see

⁵³ Note that the results by mother's education in this case include the full sample, as information is available for everyone. In the analysis of the long-term child outcomes, mother's education is only known for children who still live with their mothers.

⁵⁴ We also interact the reform effect with a continuous measure of regional unemployment, but with the same result that the coefficient on the interaction term is not statistically significant (results available upon request).

⁵⁵ Note that direction of the selection implied by the sign of the coefficients, even if not statistically significantly different from zero, suggests a complex selection mechanism with factors that are likely to offset each other. While mothers with more human capital seem to be slightly less likely to prolong their leave, suggesting negative selection, those in regions with better economic conditions (lower unemployment) are more likely to stay at home, implying positive selection. Similarly, based on the results in Online Appendix Table A.4, two very different groups of mothers — those who live with an elderly member in the household and those who have ever worked — are both more likely to select into take-up.

	Complete	ed high sch	nool		Tertiary			
	Treat1	Treat2	Treat3	Treat4	Treat1	Treat2	Treat3	Treat4
Panel A: Baseline								
Treat*After	-0.034	-0.039*	-0.030*	-0.013	0.006	-0.027	-0.053*	-0.056^{**}
	(0.030)	(0.020)	(0.017)	(0.014)	(0.053)	(0.037)	(0.030)	(0.026)
Observations	1448	2853	4254	5616	1448	2853	4254	5616
Mean of dep. var.	0.918	0.923	0.922	0.925	0.449	0.448	0.442	0.454
Panel B: By gender								
Girls	-0.063	-0.037	-0.023	-0.006	-0.065	-0.030	-0.057	-0.047
	(0.042)	(0.029)	(0.023)	(0.019)	(0.076)	(0.053)	(0.043)	(0.037)
Boys	-0.005	-0.040	-0.036	-0.020	0.076	-0.024	-0.048	-0.065*
	(0.041)	(0.028)	(0.024)	(0.020)	(0.072)	(0.051)	(0.041)	(0.036)
Observations	1448	2853	4254	5616	1448	2853	4254	5616
Mean of dep. var.	0.918	0.923	0.922	0.925	0.449	0.448	0.442	0.454
Panel C: By mother.	s' educatio	on						
Low education	-0.062	-0.042	-0.042	-0.014	-0.115	-0.073	-0.117*	*-0.126***
	(0.057)	(0.042)	(0.033)	(0.028)	(0.093)	(0.067)	(0.054)	(0.047)
High education	0.031	-0.009	-0.005	-0.003	0.024	-0.025	-0.023	-0.020
	(0.033)	(0.023)	(0.019)	(0.015)	(0.080)	(0.056)	(0.045)	(0.039)
Observations	1001	1932	2855	3796	1001	1932	2855	3796
Mean of dep. var.	0.941	0.946	0.945	0.948	0.517	0.510	0.506	0.516
Panel D: Girls by m	others' ed	ucation						
Low education	-0.002	-0.003	-0.065	-0.035	-0.301*	-0.158	-0.194*	*-0.168**
	(0.0865)	(0.053)	(0.046)	(0.040)	(0.162)	(0.108)	(0.088)	(0.077)
High education	-0.014	0.006	0.010	0.015	-0.129	-0.030	-0.047	-0.041
	(0.039)	(0.028)	(0.022)	(0.017)	(0.115)	(0.078)	(0.062)	(0.052)
Observations	442	871	1289	1747	442	871	1289	1747
Mean of dep. var.	0.957	0.960	0.960	0.962	0.654	0.651	0.649	0.651
Panel E: Boys by me	others' edi	ucation						
Low education	-0.031	-0.038	-0.035	-0.004	0.041	0.016	-0.045	-0.082
	(0.0822)	(0.061)	(0.048)	(0.041)	(0.119)	(0.086)	(0.068)	(0.059)
High education	0.075	-0.010	-0.012	-0.012	0.171	-0.051	-0.012	-0.009
	(0.051)	(0.035)	(0.029)	(0.024)	(0.121)	(0.082)	(0.067)	(0.057)
Observations	559	1061	1566	2049	559	1061	1566	2049
Mean of dep. var.	0.928	0.934	0.933	0.936	0.410	0.394	0.389	0.401

Table 3 Estimation results: education

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1 (Panel A), Treat*After interacted with dummy variables for the gender of the child (Panel B), and Treat*After interacted with dummy variables capturing the highest level of mothers' education (Panels C, D, and E). Results are presented for four different treatment and control groups (Treat1-Treat4), which differ by the size of the window around the cut-off birth date — October 1992 (see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2010-2016), own calculations

Panel A of Table 3), suggesting a decrease in the probability of enrolling in college by about 5 p.p. This is consistent with a higher intensity of treatment in the two broader specifications (due to the gradual adjustment of the new cohorts of eligible mothers of 3-year-olds), as documented in Section 4.1. The estimate is also negative, but not statistically significant, for tertiary studies for the Treat 2 specification and for the remaining two specifications for high school completion.

When we allow the impact of the reform to vary by gender (Panel B of Table 3), the results are very similar to the baseline model and across boys and girls in terms of magnitude, especially in the broader specifications, but lose statistical significance.

Estimating the impact separately for children of low- and high-educated mothers (Panel C of Table 3) reveals that the negative effect on college enrollment is driven by children with low-educated mothers, implying that the inputs to child development provided by formal childcare at the age of 3 are superior to those provided by low-educated mothers. Our results from the two broader specifications (Treat 3 and 4) suggest that the reform reduced the probability of enrolling in college by about 12 p.p. for children with low-educated mothers. The sign of the effect in the two specifications with narrower windows around the cutoff point (1 and 2 quarters) is also negative but the estimates are not statistically significant using these smaller samples.

When we conduct our heterogeneity analysis (by mother's education) also by child's gender (Panels D and E of Table 3), we find that girls are much more affected by the reform than boys. In fact, the entire negative effect of the reform on college enrollment seems to be driven predominantly by girls with low-educated mothers. The estimates suggest that the reform reduced their probability of being in or having completed tertiary education at age 21–22 by as much as 17–19 p.p. The effect is again statistically significant for the broader specifications (Treat 3 and 4) but it is now also marginally statistically significant for the narrowest window (Treat 1).

4.3 Labor market outcomes of children

We next focus on the impact on the economic status of affected children at age 21–22, as measured by the following labor market outcomes: the probability of being NEET (not in education, employment or training), employed, unemployed, and inactive with housework as main activity.⁵⁶

While we find no statistically significant impact of the 1995 reform on the probability of being employed, we do find a positive impact on being NEET (Panel A of Table 4). This is consistent with the lower probability of attending college at 21–22 (documented earlier in Section 4.2) but also reflects the impact on labor market outcomes discussed below. The positive effect on being NEET is statistically significant in all specifications except the three-quarter window (Treat 3) and suggests that the 1995 reform increased the probability of being NEET by 4–7 p.p.

The results by gender are again similar for boys and girls and close to the baseline estimate in terms of magnitude but lose statistical significance (Panel B of Table 4).

⁵⁶ All results presented in this section use the full set of control variables described in Section 3.2, but do not control for educational attainment as this is also likely to be affected by the reform (see Section 4.2). Controlling for highest level of education does not change the results (estimates available upon request).

	NEET				Being employed			
	Treat1	Treat2	Treat3	Treat4	Treat1	Treat2	Treat3	Treat4
Panel A: Baseline								
Treat*After	0.068*	0.056**	0.028	0.037**	-0.045	-0.034	0.022	0.015
	(0.039)	(0.026)	(0.021)	(0.018)	(0.053)	(0.037)	(0.030)	(0.026)
Observations	1448	2853	4254	5616	1448	2853	4254	5616
Mean of dep. var.	0.137	0.133	0.130	0.124	0.487	0.494	0.501	0.496
Panel B: By gender								
Girls	0.081	0.060	0.021	0.038	0.047	-0.007	0.054	0.036
	(0.060)	(0.041)	(0.033)	(0.028)	(0.075)	(0.052)	(0.043)	(0.037)
Boys	0.060	0.052*	0.034	0.034	-0.133*	-0.062	-0.009	-0.005
-	(0.047)	(0.031)	(0.025)	(0.021)	(0.074)	(0.052)	(0.042)	(0.037)
Observations	1448	2853	4254	5616	1448	2853	4254	5616
Mean of dep. var.	0.137	0.133	0.130	0.124	0.487	0.494	0.501	0.496
Panel C: By mothers	' education	1						
Low education	0.209***	0.133**	0.094**	0.116***	-0.196*	-0.070	0.036	0.005
	(0.075)	(0.052)	(0.042)	(0.035)	(0.102)	(0.073)	(0.058)	(0.050)
High education	0.031	0.024	0.004	0.003	0.014	-0.026	0.002	0.022
-	(0.038)	(0.030)	(0.024)	(0.020)	(0.082)	(0.057)	(0.046)	(0.040)
Observations	1001	1932	2855	3796	1001	1932	2855	3796
Mean of dep. var.	0.098	0.093	0.092	0.086	0.45	0.459	0.464	0.458
Panel D: Girls by me	others' edu	cation						
Low education	0.211*	0.150*	0.110*	0.136**	0.035	0.056	0.126	0.039
	(0.116)	(0.079)	(0.067)	(0.057)	(0.165)	(0.113)	(0.089)	(0.078)
High education	0.042	0.020	-0.012	-0.006	0.126	-0.061	-0.004	0.039
	(0.060)	(0.048)	(0.038)	(0.030)	(0.128)	(0.085)	(0.068)	(0.057)
Observations	442	871	1289	1747	442	871	1289	1747
Mean of dep. var.	0.102	0.095	0.098	0.090	0.351	0.361	0.359	0.359
Panel E: Boys by mo	thers' educ	cation						
Low education	0.184*	0.114*	0.076	0.101**	-0.355*	*-0.183*	-0.053	-0.042
	(0.099)	(0.069)	(0.055)	(0.045)	(0.137)	(0.100)	(0.079)	(0.068)
High education	0.026	0.034	0.018	0.009	-0.101	0.008	0.010	0.020
	(0.054)	(0.043)	(0.034)	(0.028)	(0.122)	(0.082)	(0.066)	(0.056)
Observations	559	1061	1566	2049	559	1061	1566	2049
Mean of dep. var.	0.095	0.091	0.088	0.083	0.530	0.540	0.550	0.542

Tab	le 4	Estimation	results:	NEET	and	empl	loyment
-----	------	------------	----------	------	-----	------	---------

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1 (Panel A), Treat*After interacted with dummy variables for the gender of the child (Panel B), and Treat*After interacted with dummy variables capturing the highest level of mothers' education (Panels C, D and E). Results are presented for four different treatment and control groups (Treat1-Treat4), which differ by the size of the window around the cut-off birth date — October 1992 (see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2010-2016), own calculations

The estimates for the other three labor market outcomes are noisier but generally consistent with the main findings.

Looking at the heterogeneity by mother's education (Panel C of Table 4), we find that the reform increased the probability of being NEET among children with loweducated mothers by 9–21 p.p., depending on the specification. While there is a fairly wide range in the estimated magnitudes, the effects are highly statistically significant for all treatment and control groups. On the other hand, the estimates for individuals with a high-educated mother are not statistically significantly different from zero in any specification, suggesting that the effect is again driven by children who were exposed to maternal care by a low-educated mother instead of attending public kindergarten.

When we further split our sample by the gender of the child, we find that the positive impact of the 1995 reform on the probability of being NEET is present for both boys and girls with low-educated mothers, with a larger effect for the latter (Panels D and E of Table 4). The results are only marginally statistically significant in the smaller samples of the narrower specifications but remain highly statistically significant for the broadest window (Treat 4). There is also some evidence that the increase in NEET among boys with low-educated mothers is driven by a decrease in their probability of being employed.

The reform had no impact on girls' employment (Panel D of Table 4), suggesting that the rise in the probability of daughters of low-educated mothers being NEET was entirely driven by the decrease in the probability of their enrolling in college evidenced above. So what were these girls doing instead of going to college? The estimates in Table 5 reveal that at least some of them were more likely to be at home doing housework.

While this is only suggestive evidence, statistically significant only for the narrowest specification (Treat 1, which is based on a small sample and corresponds to the first quarter when the effect of the treatment was limited due to the gradual adjustment of mothers to the new policy), the size of the coefficient (almost 16 p.p.) is not negligible and remains positive and around 5 p.p. in all three broader specifications. In addition to the Treat 1 specification, this is also confirmed by the Treat 4 specification for all children of low-educated mothers, regardless of child's gender, with a statistically significant coefficient and an impact of about 2.5 p.p. While the reform increased the likelihood of being NEET for both sons and daughters of low-educated mothers, we find no impact on their unemployment (see Table 5).

4.4 Family outcomes

The observed negative impact of the 1995 reform on enrollment into tertiary education and subsequent labor market outcomes could be a consequence of changes in profamily attitudes and the preferred timing of family formation. Exposure to prolonged maternal care and the potential impact of extended leave on mothers' careers and the division of gender roles in the family could stand behind the suggestive evidence of an increase in the probability of home production among girls. We therefore next explore whether the documented effects on education and labor market outcomes were accompanied by changes in marital status, cohabitation, or fertility decisions.

	Reing un	Being unemployed				Inactive - housework			
	Treat1	Treat2	Treat3	Treat4	Treat1	Treat2	Treat3	Treat4	
Panel A: Baseline									
Treat*After	0.010	0.018	0.005	0.007	0.025	0.008	0.002	0.007	
	(0.028)	(0.018)	(0.015)	(0.013)	(0.025)	(0.017)	(0.013)	(0.011)	
Observations	1448	2853	4254	5616	1448	2853	4254	5616	
Mean of dep. var.	0.066	0.062	0.064	0.061	0.053	0.052	0.051	0.048	
Panel B: By gender									
Girls	0.021	0.032	0.018	0.016	0.037	0.009	-0.006	0.008	
	(0.036)	(0.025)	(0.021)	(0.018)	(0.049)	(0.034)	(0.027)	(0.022)	
Boys	-0.001	0.004	-0.009	-0.004	0.018	0.007	0.009*	0.005	
	(0.041)	(0.027)	(0.022)	(0.019)	(0.012)	(0.007)	(0.005)	(0.004)	
Observations	1448	2853	4254	5616	1448	2853	4254	5616	
Mean of dep. var.	0.066	0.062	0.064	0.061	0.053	0.052	0.051	0.048	
Panel C: By mothers	' education	n							
Low education	0.056	0.047	0.014	0.030	0.074**	0.036	0.028	0.025*	
	(0.064)	(0.044)	(0.035)	(0.030)	(0.033)	(0.023)	(0.019)	(0.015)	
High education	-0.007	-0.003	-0.008	-0.010	0.008	-0.006	-0.007	-0.001	
	(0.033)	(0.025)	(0.021)	(0.018)	(0.014)	(0.009)	(0.007)	(0.005)	
Observations	1001	1932	2855	3796	1001	1932	2855	3796	
Mean of dep. var.	0.067	0.063	0.064	0.060	0.012	0.011	0.011	0.010	
Panel D: Girls by m	others' edu	cation							
Low education	-0.011	0.044	0.007	0.017	0.156**	0.062	0.051	0.050	
	(0.104)	(0.066)	(0.056)	(0.046)	(0.074)	(0.049)	(0.040)	(0.032)	
High education	0.032	0.013	0.000	0.002	0.008	-0.023	-0.023	-0.003	
	(0.044)	(0.034)	(0.030)	(0.026)	(0.039)	(0.025)	(0.016)	(0.012)	
Observations	442	871	1289	1747	442	871	1289	1747	
Mean of dep. var.	0.061	0.055	0.060	0.055	0.027	0.024	0.024	0.021	
Panel E: Boys by mo	others' edu	cation							
Low education	0.067	0.047	0.014	0.042	N/A	0.013	0.007	0.005	
	(0.085)	(0.061)	(0.047)	(0.040)		(0.012)	(0.007)	(0.005)	
High education	-0.032	-0.011	-0.012	-0.020	N/A	-0.000	-0.001	-0.001	
	(0.056)	(0.037)	(0.030)	(0.026)		(0.002)	(0.001)	(0.001)	
Observations	559	1061	1566	2049	559	1061	1566	2049	
Mean of dep. var.	0.072	0.069	0.067	0.064	0.000	0.001	0.001	0.000	

Table 5 Estimation results: unemployment and housework

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1 (Panel A), Treat*After interacted with dummy variables for the gender of the child (Panel B), and Treat*After interacted with dummy variables capturing the highest level of mothers' education (Panels C and D). Results are presented for four different treatment and control groups (Treat1-Treat4), which differ by the size of the window around the cut-off birth date — October 1992 (see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Due to small sample size and a very low share of boys who are inactive with housework as main activity, regressions with N/A cannot be estimated. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2010-2016), own calculations

	Treat1	Treat2	Treat3	Treat4	Treat1	Treat2	Treat3	Treat4
	Living w	ith mother			Married			
Treat*After	0.054	-0.018	-0.029	-0.016	-0.006	0.008	0.013	0.009
	(0.047)	(0.033)	(0.027)	(0.023)	(0.020)	(0.014)	(0.012)	(0.010)
Observations	1448	2853	4254	5616	1448	2853	4254	5616
Mean of dep. var.	0.715	0.702	0.696	0.702	0.037	0.040	0.040	0.039
	Having a	wn childre	en		Cohabiti	ng		
Treat*After	-0.025	-0.002	-0.003	0.010	-0.031	0.011	0.015	0.011
	(0.030)	(0.020)	(0.016)	(0.014)	(0.043)	(0.030)	(0.025)	(0.021)
Observations	1448	2853	4254	5616	1448	2853	4254	5616
Mean of dep. var.	0.081	0.076	0.075	0.072	0.217	0.224	0.228	0.222

Table 6 Estimation results: marital status and family outcomes

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1. Results are presented for four different treatment and control groups (Treat1-Treat4), which differ by the size of the window around the cut-off birth date — October 1992 (see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2010-2016), own calculations

Estimates of the effect of the 1995 reform on these outcomes, presented in Table 6, suggest that this was not the case. We find no impact of the reform on the marital status of affected children, on the probability that they have their own children, or on the likelihood of cohabitation at age 21–22. We also find no impact on the probability of a child living with its mother, which not only documents the zero impact of the reform on the probability of nest-leaving but also provides evidence that our heterogeneity analysis based on mothers' education is not based on a selected sample.

4.5 Persistence and degree completion

We now complement our findings on the impact of the 1995 reform on children's outcomes at the age of 21–22 by providing evidence on the persistence of the estimated effects at older ages. Table 7 presents the impact of the 1995 reform on the two main outcomes (tertiary education and NEET) of affected children at ages 24–25, 25–26, and 26–27, using our baseline analysis and the preferred specification Treat 4 (the one with the broadest window of 1 year on either side of the cutoff).

We focus first on education outcomes. Looking at older ages allows us to separate tertiary education enrollment and completion, as everyone could have completed their first degree by then (see Section 2.3).⁵⁷ The overall impact of the 1995 reform on tertiary education completion at the age of 24–25 is very similar to our baseline results at the age of 21–22, suggesting about a 4 p.p. reduction in the probability of completion. Interestingly, the impact on completion doubles at ages 25–26 and 26–27,

⁵⁷ Note that this is not the case in the baseline analysis, where outcomes are observed at age 21–22, when, for example, students enrolled in unstructured tertiary education programs such as medicine or law are still in the middle of their 4–6-year-long studies. See Section 2.3 for an overview of the Czech education system.

	Completed	tertiary		NEET			
Age of the affected children:	24–25	25–26	26–27	24–25	25–26	26–27	
Panel A: Baseline							
Treat*After4	-0.044*	-0.084***	-0.078***	-0.010	-0.008	-0.022	
	(0.026)	(0.027)	(0.029)	(0.020)	(0.021)	(0.024)	
Observations	4804	4538	3854	4804	4538	3854	
Mean of dep. var.	0.312	0.316	0.310	0.150	0.163	0.177	
Panel B: By gender							
Female	-0.038	-0.085^{**}	-0.083*	-0.009	-0.011	-0.034	
	(0.039)	(0.040)	(0.044)	(0.034)	(0.037)	(0.041)	
Male	-0.051	-0.082^{**}	-0.072*	-0.010	-0.004	-0.007	
	(0.035)	(0.035)	(0.038)	(0.020)	(0.020)	(0.022)	
Observations	4804	4538	3854	4804	4538	3854	
Mean of dep. var.	0.312	0.316	0.310	0.150	0.163	0.177	

Table 7 Persistence of the estimated impact

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1 at alternative ages of the treatment group. Results are presented for the largest treatment and control groups (Treat4, see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01. Source: Czech Labor Force Survey data (2014–2020), own calculations

indicating that affected children were not only less likely to enroll but also more likely to drop out without completing their studies than those not affected by the reform.⁵⁸

Regarding NEET, the results suggest that the impact of the 1995 reform was most pronounced among recent high school graduates at age 21–22, as documented in our baseline analysis, but disappeared at later ages. This is consistent with the difficulties that high school graduates typically face when entering the labor market for the first time. However, the difference in educational attainment and the impact of the temporary absence from the labor market after high school for those not enrolled are likely to have substantial negative effects on lifetime earnings as well as job quality, implying a persistent impact of the 1995 reform on the well-being of affected children. Table 7 also presents the results by gender. The impact of the 1995 reform on tertiary education completion is confirmed and the size of the effect is similar for girls and boys.

The results of the heterogeneity analysis (available upon request) are qualitatively similar to our main findings, confirming that the impact on tertiary education completion is driven primarily by children of low-educated mothers, but they are much noisier because the sample size of children who still live with their mothers at older ages, and thus have information about their mother's education, declines sharply with age (to less than 10% at age 26–27).

 $^{^{58}}$ Note that the reduction in sample size at older ages is due to the overall decrease in the number of respondents in the most recent surveys of the LFS data.

5 Identification assumptions and robustness checks

5.1 Alternative specification

In our main specification, we compare the last-unaffected and first-affected cohorts (born before and after October 1992) at exactly the same age (measured in quarters of a year) when they are 21–22 years old. In order to filter out calendar time and quarter-ofbirth effects when comparing cohort outcomes, we use the difference-in-differences approach with control cohorts of older children (born before and after October 1990), observed over exactly the same calendar time and with the same quarter-of-birth compositions. Our identification strategy is thus based on the assumption that the potential confounding factors are the same for the treatment and control cohorts observed in the same calendar year but at different ages.

To assess the sensitivity of our results to these assumptions, we conduct a robustness check that compares the outcomes of the last-unaffected and first-affected cohorts in exactly the same calendar time (but at different ages), controlling for potential age and quarter-of-birth effects using the control cohorts (born before and after October 1990) observed at the same age. Summary statistics of the treatment and control cohorts in this alternative specification can be found in Online Appendix Table A.5.

While this alternative specification allows for the age-specific impact of changes in economic conditions and quarter-of-birth effects, it makes a different restrictive identification assumption that age and quarter-of-birth effects do not vary with calendar time. In particular, outcomes are allowed to be affected by age and/or quarter-of-birth effects, but the impact of age and quarter-of-birth on these outcomes must be timeinvariant. To provide an example of a violation of these assumptions, if the probability of employment changes differently with age in times of recession than in times of boom, and we observe our treated and control cohorts at different stages of the business cycle, the control group will not fully filter out age effects.

To explore the robustness of our results and to show how they vary with the various assumptions imposed, we next compare the results from the two specifications. The estimates from the main specification were presented in Tables 3, 4 and 5 in Section 4. The corresponding results using the alternative specification are reported in Tables A.6, A.7 and A.8 in the Online Appendix. Although the robustness analysis shows no effect of the 1995 reform on the tertiary-education outcome (Online Appendix Table A.6), it confirms the large and statistically significant effect of over 10 p.p. on the NEET outcome for girls with low-educated mothers (Panel D in Online Appendix Table A.7). We see similar but stronger evidence that the reform increased the likelihood of home production among girls with low-educated mothers (Panels C and D in Table A.8 in the Online Appendix). In particular, the reform's impact on being inactive with housework as main activity among girls with low-educated mothers is highly statistically significant in all but the narrowest specification (Treat 1), suggesting an increase of 7–9 p.p.

Interestingly, the robustness results suggest that the reform reduced the probability of being unemployed among boys by about 4.5 p.p. (Panel B in Online Appendix Table A.8). This is the first finding that implies some positive effect of the reform on the labor market outcomes of affected children. While this finding is not supported by

the results from our main specification, it is in line with the conclusion that the impact of the reform depended to a large extent on the mother's educational attainment (a proxy for the quality of maternal care that substituted kindergarten attendance for the affected cohorts). Panel C of Table A.8 reveals that this negative effect on unemployment was mainly driven by children of high-educated mothers and Panel E again confirms that the effect is driven by boys. Since there is no impact on the probability of being NEET or in employment, the decrease in unemployment suggests that the sons of (mostly) high-educated mothers continued their studies, but not at university, as there is no evidence of an increase in enrollment in tertiary education.⁵⁹ Our findings on the impact of the 1995 reform on the probability of living with one's mother, on partnership status and on family formation (considered in Section 6) are all robust to the alternative specification, confirming that the extension of family leave had no impact on any of these outcomes (Table A.9 in the Online Appendix).

5.2 Common trend assumptions

We provide further evidence on the validity of these two estimation strategies by checking their underlying common trend assumptions. Our approach assumes that the evolution of the outcome variables for the treatment and control cohorts would have been the same in the absence of the reform. We check this assumption by comparing the series of cohorts born in the pre-treatment period and thus not affected by the reform. Figure A.1 in the Online Appendix shows pre-reform cohort trends for the main specification, where the corresponding treatment and control cohorts are observed in the same calendar time (the year 2014). The figures show averages of our main outcome variables for the treatment cohorts born before October 1, 1992, and for the control cohorts born before October 1, 1990 (cohorts born one quarter before the corresponding October threshold are denoted -1, those born two quarters before the corresponding threshold are denoted -2, and so on). Both the education and labor market outcomes of the treatment and control cohorts evolve more or less in parallel over the pre-treatment period, with the exception of cohorts -2, for which the share of individuals in tertiary education is higher and the share of NEET individuals is lower in the control cohort (relative to the treated cohort) compared to what the common trend would have implied.

Figure A.2 in the Online Appendix illustrates the pre-treatment trends for the alternative specification by comparing the pre-reform evolution of the outcome variables for the treated and control cohorts observed at the same age (21). Most of our outcome variables are slightly more volatile in the treatment cohorts than in the control cohorts, so the trends are less similar than in the main specification. There is, however, one outcome variable for which the common trend seems to work better in the robustness specification - the share of individuals being inactive with housework as main activity. Recall that while the coefficients in the robustness results are not statistically significant for the education outcomes, they show much stronger evidence of a reform's

⁵⁹ In the broadest specification (Treat 4), we also find a negative effect of the reform on unemployment for boys with low-educated mothers, but this is not confirmed by the narrower specifications.

effect on the probability of doing housework than we found in our main specification. There is a trade-off in choosing between the main and alternative specifications, as each imposes a different set of potentially non-innocuous identification assumptions. Neither approach can be considered superior and their appropriateness for a particular empirical question also depends on the outcomes being analyzed. Since we are studying two types of outcomes, different assumptions may be more likely to be violated for one type than for the other: While education outcomes are likely to be less affected by current macroeconomic conditions (captured by calendar year effects), they may be more sensitive to the exact age (captured by age effects). On the other hand, comparing labor market outcomes at the same calendar time but at slightly different ages seems to be more innocuous than doing so at exactly the same age but in different economic conditions (as reflected by calendar time). This was also confirmed by the common trend assumptions of the two alternative specifications for the two types of outcomes discussed above. Accordingly, we conjecture that our main specification may be more suitable for education outcomes, while the alternative specification may be more appropriate for studying labor market effects. Although they differ in the exact magnitude and statistical significance of the estimates, they are unanimous in documenting the negative impact of the 1995 reform on the long-term outcomes of children of low-educated mothers.

5.3 Cohort outcomes' evolution around the reform

The standard approach in the literature to graphically examine the impact of reform is to construct event study graphs (for details see, e.g., Clarke and Tapia-Schythe 2021) that show the evolution of outcomes for the treatment and control groups over a pre-treatment and post-treatment period. Unfortunately, there is no such control group in our design. Since all children born after October 1, 1992, were affected by the reform, we use cohorts born around a similar cut-off date of October 1, 1990, i.e., 2 years before the treated cohorts, as control cohorts. As a result, we are unable to plot standard event study graphs because our pre- and post-treatment periods consist of only 1 year (or 4 quarters) each. If we extend the post-treatment period beyond 1 year, the control cohorts begin to overlap with the cohorts in our treatment group.

In order to provide at least some evidence on the evolution of the two main outcome variables around the reform, Figs. A.3 and A.4 in the Online Appendix plot the shares of those enrolled in tertiary education and the shares of NEET for cohorts born up to 4 years before and 6 years after the reform (relative to the last-unaffected cohort, i.e., those born in 1991). The evolution of the share in tertiary education mainly reflects the upward trend in college enrollment driven by the long-term expansion of tertiary education in the Czech Republic over the last three decades. The slightly decreasing trend in the probability of being NEET over time may be related to the gradual transformation of the communist education system, which had an increasingly positive effect on the quality of skills acquisition of each new cohort, thus improving their future labor market outcomes. While there is no clear break in these long-term trends in 1992, there is a noticeable slowdown in the evolution of the share of tertiary

educated (Fig. A.3 in the Online Appendix) and a clear upward shift in the evolution of the share of NEET (Fig. A.4 in the Online Appendix) in the 2 years after the reform. Both are consistent with our findings of a negative impact of the reform on tertiary education and a positive impact on NEET.

Based on previous research and our own results, we would expect the reform to have a larger impact on the children of less-educated mothers. Figures A.5 and A.6 in the Online Appendix replicate the evolution of the two main child outcomes by mother's education. To the extent that the impact of the reform was weaker for those with mothers with higher levels of human capital, the cohorts of children of high-educated mothers can be seen as a "pseudo-control" group for the cohorts of children of low-educated mothers and the differences between the two groups can be interpreted as suggestive evidence of the potential impact of the reform on the latter relative to the former.

Figures A.5 and A.6 in the Online Appendix confirm the diverse evolution of the outcomes for the two groups. There is a drop in the share of individuals in tertiary education in the post-reform period for cohorts of children with low-educated mothers, while no such decrease is observed for cohorts with high-educated mothers. Similarly, the share of individuals in NEET increases after the reform for cohorts with low-educated mothers, but not for cohorts with highly educated mothers. These differences are consistent with our baseline results on the impact of the 1995 reform on child long-term outcomes and with the conclusion of our heterogeneity analysis that the estimated impacts are mostly driven by children of low-educated mothers.

5.4 Alternative control groups

In our baseline analysis, we use the cohorts born before/after October 1, 1990, as a control group for the cohorts born before/after October 1, 1992, who were the lastunaffected and the first-affected cohorts by the 1995 reform. As a robustness check of our results, we re-estimate our baseline analysis using older cohorts, namely those born around October 1, 1988, and around October 1, 1989. The estimates of the impact of the 1995 reform using these alternative control cohorts for the two main outcomes (enrollment in tertiary education and NEET) for our preferred specification Treat 4 are presented in Table 8. The results using the control group of cohorts born around 1989 are quite similar to our baseline analysis, with a statistically significant reduction of 4.3 p.p. (compared to the baseline estimate of 5.6 p.p.) in tertiary education enrollment and an increase of 4.3 p.p. (compared to the baseline estimate of 3.7 p.p.) in the probability of being NEET. The heterogeneity analysis also confirms our main findings in terms of magnitude but the education outcome for children of low-educated mothers loses statistical significance. Using even older cohorts, those born around 1988, our findings are no longer supported. We attribute this in part to the fact that the older cohorts (born 1 year before the end of the communist regime in the Czech Republic in November 1989) are simply too different from cohorts born around 1992. More importantly, the outcomes of our treatment and control groups are measured at different ages (to control for the same calendar year and quarter-of-birth effects). In our baseline analysis, the control group cohorts are observed at age 23–24, while the treatment group cohorts

Table 8 Robustness check: alternative control cohorts		Control co born arour	Control cohorts born around 1989		Control cohorts born around 1988				
		Tertiary	NEET	Tertiary	NEET				
	Panel A: Baseline								
	Treat4*After	-0.043*	0.043**	-0.010	0.016				
		(0.026)	(0.018)	(0.026)	(0.019)				
	Observations	5637	5638	5554	5555				
	Mean of dep. var	0.434	0.137	0.423	0.143				
	Panel B: By mothers' education								
	Low education	-0.065	0.096***	-0.033	0.033				
		(0.048)	(0.037)	(0.050)	(0.039)				
	High education	-0.036	-0.001	0.077*	-0.043*				
		(0.040)	(0.023)	(0.043)	(0.025)				
	Observations	3616	3616	3323	3323				
	Mean of dep. var	0.507	0.097	0.506	0.096				

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1 but with alternative control groups born around October 1989 and October 1988. Results are presented for the largest treatment and control groups (Treat4, see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2010-2016), own calculations

are observed at age 21–22. The alternative control group cohorts born around 1989 are observed at age 24–25, and the cohorts born around 1988 are observed at age 25–26, making their outcomes even more distant over the life cycle and therefore less comparable.

5.5 Placebo analysis

This section includes two placebo-type tests to check the internal validity of our results. First, we present a test of the common trend assumptions by conducting a placebo analysis of the impact of the 1995 reform on "fake" outcomes (those that the reform could not affect) and checking that it is zero Gertler et al. (2016). In particular, we estimate our baseline model with predetermined variables, namely mother's and father's education, as outcomes that could not have been affected by the extension of paid family leave. Table 9 confirms zero impact of the reform on these variables for all four specifications (Treat1 - Treat4), suggesting that the difference in the composition of parental education between the pre- and post-1990 cohorts in our control group is the same as that between the pre- and post-1992 cohorts in our treatment group, as described in Fig. 5.

The second test exploits a feature of our institutional setting that helps us identify a subset of treated children for whom the estimated impact of the reform should be zero

	Mother h	Mother high education				Father high education			
	Treat1	Treat2	Treat3	Treat4	Treat1	Treat2	Treat3	Treat4	
Treat*After	0.022	-0.030	-0.040	-0.042	0.069	-0.042	-0.050	-0.057	
	(0.064)	(0.045)	(0.037)	(0.032)	(0.073)	(0.050)	(0.041)	(0.035)	
Observations	1001	1933	2856	3797	847	1650	2426	3243	
Mean of dep. var.	0.606	0.615	0.603	0.607	0.510	0.511	0.500	0.500	

 Table 9
 Robustness check: fake outcomes

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1. Results are presented for four different treatment and control groups (Treat1-Treat4), which differ by the size of the window around the cut-off birth date — October 1992 (see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2010-2016), own calculations

and thus can be used as a placebo group. While all mothers of children younger than 3 were eligible to extend their family leave from 3 to 4 years when the 1995 reform came into effect on October 1, children with younger siblings were only affected through their younger siblings. As the entitlement to parental allowance and the restriction on the use of formal childcare only affected the youngest child in the family, children who had younger siblings and turned 3 before October 1 were affected in the same way as those who turned 3 after that date. Their mothers became all entitled to parental leave extension up to the 4th birthday o their younger siblings and under the same conditions: the mother was restricted from working and the younger sibling was restricted from attending formal childcare. There is no reason why the effect of the potential extension of maternal care for their younger siblings to 4 years should differ between these two cohorts or why children who turned 3 after October 1 should be more likely to have their kindergarten enrollment postponed. The impact of the 1995 reform on the longterm outcomes of children with younger siblings in our treatment group should be therefore zero since they were affected in the same way regardless of whether they turned 3 in the before or after period. We use this fact as the basis for our placebo test.

Unfortunately, the data do not allow us to accurately identify children in our treatment group who had a younger sibling at the time of the reform. We only observe siblings if both the children and their siblings are still living with their mother at the time we measure their long-term outcomes. Therefore, to avoid the possible bias due to selective nest-leaving, we use all children who turned 3 before and after the reform (regardless of the presence of a sibling) in our baseline estimation in Section 4. As an identification check, however, we can estimate the impact of the reform on a subset of children who still live with their parents and a sibling less than 3 years younger when they are 21–22. Since they were all affected by the 1995 reform in the same way — only through their younger siblings — the long-term outcomes of children in this subgroup who turned 3 before or after October 1 should not differ. The results of this placebo test, presented in Panel A of Table 10, confirm that, in contrast to our baseline analysis, there is indeed zero impact of the 1995 reform in this group, suggesting no

	Tertiary				NEET			
	Treat1	Treat2	Treat3	Treat4	Treat1	Treat2	Treat3	Treat4
Panel A: Placebo g	group							
Treat*After	-0.162	-0.018	0.000	0.032	0.044	0.025	0.012	0.061
	(0.258)	(0.170)	(0.122)	(0.101)	(0.135)	(0.075)	(0.055)	(0.051)
Observations	141	254	367	482	141	254	367	482
Mean of dep. var.	0.518	0.528	0.531	0.519	0.092	0.075	0.063	0.071
Panel B: Main sam	ple exclud	ling place	bo group					
Treat*After	-0.007	-0.060	-0.081^{**}	-0.089^{***}	0.087**	0.060**	0.041*	0.042**
	(0.070)	(0.049)	(0.040)	(0.034)	(0.042)	(0.030)	(0.024)	(0.020)
Observations	860	1679	2489	3315	860	1679	2489	3315
Mean of dep. var.	0.560	0.511	0.501	0.519	0.109	0.096	0.090	0.086

Table 10 Placebo analysis

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1. Results are presented for four different treatment and control groups (Treat1-Treat4), which differ by the size of the window around the cut-off birth date — October 1992 (see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2010-2016), own calculations

difference in the outcomes of children who turned 3 before and after October 1, 1995, and had a younger sibling.

In contrast, when we exclude children who had a younger sibling at the time of the reform from our main sample, the results of our baseline specifications become even stronger, both in terms of statistical significance and magnitude (Panel B of Table 10).⁶⁰ In particular, the negative impact of family leave extension on the probability of enrolling into tertiary education rises from 5 p.p. to 8–9 p.p. in the broader specifications (Treat 3 and 4). The magnitude of the increase in the probability of being NEET at age 21–22 in response to the 1995 reform also rises, from 4 p.p. to 6–9 p.p., in the two narrower specifications (Treat 1 and 2) and remains similar in magnitude in the two broader specifications (Treat 3 and 4).

5.6 Sample selection in the heterogeneity analysis

As explained in Section 3.1, our heterogeneity analysis, which examines the effect of the reform by mother's education, is conducted on a restricted sample of individuals who still live with their mother (for others, mother's education is unknown). While this subsample represents as much as 75% of our main sample, the estimates may still

⁶⁰ This may indicate that our baseline estimates are only a lower bound on the true effects.

suffer from sample selection bias if the decision to leave the parental home is affected by the reform.⁶¹

However, we find no effect of the reform on the probability of children aged 21-22 living in the same household as their mothers when we estimate the effect on family outcomes in Section 4.4,⁶² suggesting that the results of our heterogeneity analysis should not be affected by selection into nest-leaving.⁶³

6 Mechanisms

6.1 Extended maternal care and postponed preschool

In this section, we compare our results with previous findings and discuss the various mechanisms that could account for the negative impact of the extension of paid family leave on child long-term outcomes implied by our estimates. We start with the most obvious, exposure to full-time maternal care at age 3 and postponement of preschool enrollment, but we also discuss other factors that could lead to the same results.

Our results are consistent with the negative impact of family leave extension to 3 years on children's schooling outcomes documented in Dustmann and Schönberg (2012) and Canaan (2022) in terms of sign but not magnitude. The effect of maternal care at age 3 on college enrollment that we find is about 20 times larger than the effect of maternal care at age 2 on the probability of attending an academic track in high school, the most comparable outcome to ours estimated in Dustmann and Schönberg (2012).⁶⁴ Comparing the effect in percentages rather than percentage points, our implied decrease in college enrollment of about 13% is still almost ten times larger than the 1.5% decrease in academic track attendance that they find. Three main factors are likely to account for the differences in magnitude: the different size of the take-up (of 25 p.p. in our case and less than 10 p.p. in theirs), the different impact of full-time maternal care at age 3 in our case and at age 2 in theirs, and the different counterfactual. The counterfactual in our setup is likely to be much more homogeneous than in previous research from settings with various alternative forms of childcare, both public, private, and other informal, as in Dustmann and Schönberg (2012). Care provided in public kindergartens in the Czech Republic was by far the most dominant alternative to maternal care with very little variation in content and quality, as documented in Section 2.2. Danzer et al. (2022) is the only study we are aware of that estimates the impact of family leave separately for different alternative types of childcare. Focusing

 $^{^{61}}$ Note that the share of those living with their mother is even lower (about 64%, see Online Appendix Table A.1) for the control cohorts in our main specification, which uses their outcomes when they are older (23–24), to compare the treatment and control cohorts at the same calendar time. This is not the case in the alternative specification in our robustness analysis, where the treatment and control cohorts are compared in different calendar years but at the same age (21–22).

⁶² This is also confirmed for the alternative specifications (Appendix Table A.9).

 $^{^{63}}$ We confirm this by rerunning our baseline specification on the subset of individuals used in the heterogeneity analysis (those living with their mother) and obtain similar results to those for the full sample (results available upon request).

⁶⁴ Because the academic track channels high school students toward college education, it can be viewed as a proxy for college enrollment.

on similar outcomes as we do, they find no impact of extending paid family leave to 2 years for children who are most comparable to our sample in terms of the counterfactual type of care, i.e., the relatively small group in their data for whom prolonged maternal care most likely replaced formal childcare provided in nurseries.⁶⁵

However, the contrast between our results and previous findings should not be seen as contradictory, but rather as documenting different pieces of evidence that complement each other. The effect of full-time maternal care versus formal childcare cannot be the same at age 1 or 2 as it is at age 3.

This is confirmed by studies evaluating universal childcare reforms that replace maternal (or informal) care with formal childcare. This strand of research documents that the importance of formal childcare attendance increases with the age of the child, especially for children of low-educated parents. The setup of our analysis is similar to that in this literature, and our results are strikingly similar to those in Havnes and Mogstad (2011), a study that is most comparable to ours. They analyze a large-scale expansion of subsidized childcare for 3–6-year-olds on their education and labor market outcomes in adulthood and find that enrollment in subsidized childcare increases the probability of attending college by almost 7 p.p. (19% increase) and reduces the probability of being on welfare by almost 5 p.p. (31% increase); magnitudes and findings very similar to ours.⁶⁶

Our study complements their results by showing that there is a positive impact of preschool education on children at age 3 not only relative to informal care, which is the dominant counterfactual in Havnes and Mogstad (2011), but also relative to care provided by the child's mother. The results of our heterogeneity analysis are also consistent with findings from this strand of the literature suggesting that the positive long-term effects of formal childcare are mostly driven by children from low socioeconomic backgrounds. We also find a larger impact for daughters, but there are slight differences in results by gender for certain outcomes. Havnes and Mogstad (2011) find that the increase in educational attainment was most pronounced for children from low socio-economic backgrounds, while improvements in labor market outcomes were mainly experienced by girls. Using the same reform, Havnes and Mogstad (2015) further explore the nonlinearity of the impact of preschool childcare and show that the positive effect is driven by individuals from the lower and middle parts of the earnings distribution, which often include less-educated parents. Finally, Felfe et al. (2015) show that the positive effects of universal childcare are driven by children from disadvantaged families and girls. While our estimates confirm a higher risk of being NEET at age 21–22 for daughters of low-educated mothers (driven by a decrease in college enrollment and a slight increase in home production), it is sons of low-educated mothers who suffer a decrease in the probability of employment, which explains the rise in their probability of being NEET.

For children who participated in Headstart (a means-tested public preschool program for 3–5-year-olds), Bailey et al. (2021) find an 8.5% increase in college enrollment. They also find that these children are 4.5 p.p. more likely to work as

⁶⁵ They proxy the use of public care with the local presence of a nursery.

⁶⁶ In percentage terms, we estimate a 13% decrease in college enrollment and over 40% increase in being NEET.

adults. They estimate the take-up rate to be about 15 p.p. Because the program was means-tested and therefore used mostly by children from lower socio-economic backgrounds, their results should be compared primarily with our results for children of low-educated mothers. Kawarazaki (2023) finds an even larger positive impact of public childcare attendance for Japan, suggesting that enrolling in formal childcare at age 4 increases children's future income by about 44% and raises the college completion rate by 38 p.p. The large effect is entirely driven by girls, and the magnitude is partly explained by the initial very low levels of female enrollment in higher education and labor force participation.

We attribute an important part of the negative impact of extended maternal care on human capital accumulation and labor market outcomes of the affected cohorts that we find to the direct effect of extended leave crowding out preschool education at age 3. Based on our findings, and consistent with the most comparable studies in the universal childcare literature, we conclude that full-time exposure to maternal care at age 3 may not provide children, especially those with low-educated mothers, with the same amount and quality of early childhood inputs that they would have received in subsidized formal childcare in the absence of the reform. In particular, the intellectual stimuli and social interactions provided by preschool education in public kindergartens at this age, which are crucial for the accumulation of cognitive and non-cognitive skills, may not be easily substitutable, especially by low-educated mothers (Broberg et al. 1997; García et al. 2020; Loeb et al. 2007).

6.2 Alternative mechanisms

Next, we discuss alternative mechanisms that could underlie our findings. The first is a possible change in the quality of formal childcare due to the reduction in preschool enrollment of 3-year-olds after the 1995 reform (see, e.g., Blanden et al. 2021 for previous evidence on the effect of childcare quality on child education outcomes). As the average age in kindergartens increased, greater exposure to older peers may have accelerated the development of enrolled children and positively affected their long-term outcomes. However, such an effect is likely to be limited because kindergartens typically group children by age (as discussed in Section 2.2), so 3-year-olds were not interacting much with older peers anyway. The decline in the number of enrolled 3-year-olds would therefore mostly be reflected in either a reduction in the number of classes with 3-year-olds or a decrease in their class size.

If children entering kindergarten at age 3 after the reform experienced lower childteacher ratios, they would be exposed to higher-quality care, which could affect their long-term outcomes. However, the average class size and the child-teacher ratio in public kindergartens do not show any noticeable changes around the 1995 reform: the average class size decreased from 22.9 to 22 and the average child-teacher ratio decreased from 12.2 to 11.7 between 1994 and 1996 (Kuchařová et al. 2009). As noted in Section 2.2, kindergarten enrollment was also significantly affected by two other factors at the time, the secular decline in fertility and the gradual reduction in the number of kindergartens due to their closure, both of which also affected these two measures. As part of the latter trend, several smaller classes were often combined into a single class, thus maintaining the same quality in terms of number of children per teacher. Most importantly, an (expected) increase in the quality of childcare provided in public kindergartens in 1995 (for which we have no evidence) would only suggest that the negative impact of the extension of family leave on child outcomes we find represents a lower bound on the true effect.⁶⁷

Finally, we consider alternative channels through which extended maternal care may affect long-term child outcomes, such as gender norms (Kleven et al. 2019), household specialization (Canaan 2022), family stability (Cygan-Rehm et al. 2018), or household income (Nicoletti et al. 2023).

The fact that girls from lower socio-economic backgrounds invest less in their human capital and focus more on home production when exposed to full-time maternal care at age 3 could be the result of a change in preferences and attitudes induced by the reform. In particular, exposure to extended maternal care could lead to a rise in pro-family attitudes, reinforce traditional gender role models, and increase children's preference for early family formation. The heterogeneity in the effect we find across genders is consistent with Kleven et al. (2019), who show that traditional gender roles are transmitted from parents to daughters but not to sons.⁶⁸ The impact of gender norms on labor market outcomes can be substantial, with magnitudes comparable to those we find here, as documented in Steinhauer (2018).⁶⁹

If extended leave changed pro-family preferences and gender norms, we would expect it to have also affected the timing of family formation and fertility, possibly substituting for human capital investment and labor market attachment, and thus explaining the adverse impact of the 1995 reform on the probability of college enrollment and NEET. However, the results already presented in Section 4.4 show no impact on nest-leaving, marital status, the probability of having own children, or the likelihood of cohabitation at age 21–22.

Next, we consider the potential impact of the 1995 reform on mothers' and family outcomes, which in turn may affect children's development. The take-up of extended family leave had a substantial impact on household income. A couple with a child aged 3 would have earned on average about 21,700 CZK in 1996 (an average male wage of about 12,200 CZK and an average female wage of about 9,500 CZK) if the mother had not extended her parental leave from the child's 3rd to 4th birthday.⁷⁰ Since the cost of sending the child to a public kindergarten was rather low (at most 1,000 CZK per month, including food), the monthly household income would be about 20,700 CZK in that year. If the mother decided to extend her parental leave by one year, the

⁶⁷ Our ITT estimates would capture a combination of a negative impact of exposure to maternal care instead of preschool education for those who postponed kindergarten enrollment, and a positive impact of exposure to higher-quality formal childcare for those whose mothers did not extend their family leave.

⁶⁸ Kleven et al. (2019) find that daughters of mothers whose wages are affected by a larger child penalty are more likely to also experience a sizable child penalty.

 $^{^{69}}$ He shows that women born into environments with strong beliefs that children of working mothers suffer have a lower probability of working when their children are very young, by as much as 5–10 p.p. (about 15–25% less than women born into social environments with less traditional beliefs).

⁷⁰ Source: Czech Statistical Office, https://www.czso.cz/documents/10180/35067255/1100251543.xlsx/ ddbbed66-7edd-4e77-8b59-9c1401fd1846?version=1.1.

household income would fall to about 14,200 CZK, which corresponds to the average male wage (12,200 CZK) plus the monthly parental allowance (1,980 CZK). The total monthly household income of families with mothers who extended their leave until the child's 4th birthday thus decreased by more than 25%.⁷¹

This drop in household income probably represents a lower bound on the total loss, as some mothers either extended their parental leave beyond the child's 4th birthday or were unable to find a job when it ended (Bičáková and Kalíšková 2019). Dahl and Lochner (2012) and Clark-Kauffman et al. (2003) document a positive effect of an increase in household income on child achievement, especially for children from disadvantaged families and those who experience the increase early in life. Nicoletti et al. (2023) shows that the positive income effect of an increase in hours worked by mothers of preschool children fully compensates for the negative effect of the reduction in their time investment on child education outcomes. The reduction in family resources associated with the additional year of parental leave may have negatively affected children's development at this age and contributed to the negative effects of the 1995 reform on child long-term education and labor market outcomes that we find. Mothers' prolonged absence from the labor market could have further negative effects on household resources and family background due to their slower career progression and lower future earnings or job loss.⁷² Extending paid family leave could also have an impact on family stability, affecting both household income and children's well-being Cygan-Rehm et al. (2018).

We do not have information in the data on mothers' earnings, post-leave job quality, or household income, but we can explore a subset of other maternal outcomes to shed more light on the potential impact of the reform on long-term household earnings and family stability. In particular, we estimate the effect of the 1995 reform on mother's economic outcomes (probability of employment/unemployment) and on their probability of being single and divorced at different points in time during the affected children's early and late adolescence, specifically at 2-year age windows between ages 10 and 17.⁷³ These estimates should not only capture any impact of the 1995 reform on mothers' medium-term outcomes (about a decade after the possible extension of their parental leave) but also document any differences in these outcomes at the most formative and vulnerable ages of the affected children.⁷⁴

⁷¹ There is no evidence that husbands try to mitigate this income loss by increasing their labor supply, as previously documented, for example, in (Ginja et al. 2023) or (Canaan 2022). Bičáková et al. (2011) find zero elasticity of male labor supply with respect to spousal employment in the Czech Republic. The majority of men worked full time at that time, constrained by the institutional setting of rather rigid contractual arrangements, a legacy of the communist era.

⁷² Pertold-Gebicka (2020), for example, who evaluates another reform of parental leave in the Czech Republic in 2008, shows that shortening parental leave leads to more women working in high-skilled occupations 6 to 8 years after childbirth.

 $^{^{73}}$ Note that mothers' outcomes are observed for the vast majority of children at these ages because they still live with their mothers (95% of children live with their mothers at age 17).

⁷⁴ Hoxby (2021) identifies ages 10–15 for girls and 11–16 for boys as the most important for the development of advanced cognitive skills.

Age of the affected children:	10–11	12–13	14–15	16–17	10–11	12–13	14–15	16–17
	Mother e	mployed			Mother 1	unemployed	d	
Treat4*After	0.126	-0.013	-0.051	0.014	0.008	0.012	0.028	0.006
	(0.102)	(0.050)	(0.047)	(0.047)	(0.066)	(0.032)	(0.027)	(0.025)
Observations	3380	4670	4837	4748	3380	4670	4837	4748
Mean of dep. var.	0.679	0.745	0.766	0.789	0.092	0.081	0.073	0.058
	Mother s	ingle			Mother divorced			
Treat4*After	-0.057	-0.024	-0.019	-0.020	0.016	0.006	-0.047	-0.004
	(0.074)	(0.038)	(0.042)	(0.046)	(0.058)	(0.035)	(0.041)	(0.044)
Observations	3180	4461	4626	4499	3380	4670	4837	4748
Mean of dep. var.	0.124	0.141	0.155	0.166	0.108	0.125	0.157	0.163

Table 11 Estimation results: mother's outcomes when children are aged 10-17

Note: The table reports estimated coefficients of the interaction term Treat*After from equation 1. Results are presented for the largest treatment and control groups (Treat4, see Online Appendix Table A.2 for details). All regressions control for a full set of control variables. Standard errors are in parentheses. *p < 0.1; **p < 0.05; ***p < 0.01.

Source: Czech Labor Force Survey data (2005-2011), own calculations

The results summarized in Table 11 show no evidence that the reform had an impact on mothers' long-term labor market outcomes or marital status at any point during the early and late adolescence of the affected children.⁷⁵

While we cannot rule out the presence of effects of the reform on other maternal and family outcomes that we do not observe in our data (including household income, mothers' earnings or post-leave job quality) or the impact of the reform on children's values and aspirations, the limited evidence we have points to exposure to maternal care rather than attendance at subsidized kindergarten between ages 3 and 4 as the most important mechanism behind our findings.

7 Conclusion

Overly generous family leave policies may have substantial unintended negative consequences for mothers and children in the short run, according to a number of recent studies. Extended maternal care of several years has been documented to have a negative impact on children's schooling outcomes (Dustmann and Schönberg 2012 and Canaan 2022) but the evidence on the long-term effect has been lacking. This paper shows that the negative effects are persistent and translate into poorer education and labor market outcomes in early adulthood.

Using the 1995 parental leave reform in the Czech Republic, which extended paid family leave until the child's 4th birthday, we show that full-time maternal care at the

⁷⁵ This is consistent with Bičáková and Kalíšková (2019), who document that the negative effects of the 1995 reform on mothers' immediate labor market outcomes dissipated by the time the youngest child turned 7.

age of 3 reduces college enrollment and labor market attachment at the age of 21–22. The negative impact on college enrollment is driven by children of low-educated parents and is larger for girls than for boys. Focusing on college completion at later ages, we find that the negative impact on enrollment reduces the probability of completing tertiary education by about 8 p.p. for both genders in the affected cohorts. The risk of being neither in education nor in employment at the age of 21–22 increases by as much as a 12 p.p. for children of low-educated parents, again with a somewhat larger impact on girls than on boys. There is some evidence that different factors are at play behind this effect for different genders. Daughters of low-educated mothers are more likely to be inactive with housework as main activity, while sons are less likely to be employed. In contrast to the effect on education, the increase in the probability of being neither in education nor in employment seems to be short-lived. Finally, we find no effect of the reform on nest-leaving, family formation, or fertility at the age of 21–22.

The paid family leave extension had universal eligibility and induced an additional 27% of mothers of 3-year-olds to prolong their maternal care beyond their child's 3rd birthday. This is a much higher take-up rate than has been observed in previous studies of family leave reforms. While we cannot rule out the possibility that these mothers differed from those who did not take up the leave (or who would have stayed at home until the child's 4th birthday anyway) in terms of unobservable factors, we find no evidence of selectivity in take-up based on mothers' observable characteristics or pre-reform labor market conditions.

Our findings are robust to comparing the long-term outcomes of children in our treatment and control groups at the same age or at the same calendar time, as well as to using older cohorts as alternative control groups. Two placebo tests also confirm the validity of our results.

We provide suggestive evidence that the extended maternal care induced by the 1995 reform postponed the enrollment of the affected children in public kindergartens by 1 year. We therefore attribute an important part of the negative impact of the 1995 reform on human capital accumulation and labor market outcomes of the affected cohorts to the direct effect of the extended leave crowding out preschool education. The estimated effects are comparable to those found in research evaluating the impact of universal childcare at the age of 3 on long-term child outcomes. On the basis of our findings and the conclusions drawn from this strand of research, we conjecture that the quantity and quality of early childhood inputs provided by full-time maternal care may be inferior to what the affected children would have received in subsidized formal childcare in the absence of the reform. In particular, the intellectual stimuli and social interactions provided as part of preschool education in public kindergartens at this age, which are crucial for the accumulation of cognitive and non-cognitive skills, may not be easily substituted by maternal care, especially by less-educated mothers (Broberg et al. 1997; García et al. 2020; Loeb et al. 2007).

However, other mechanisms may also be at work. In particular, the reduction in inputs could also be driven or amplified by the overall drop in total household income as a result of mother's foregone earnings during the additional year of parental leave. The larger negative impact on human capital investment and labor force attachment of daughters of low-educated mothers, who instead seem to focus more on home production, may also suggest a potential impact of the extended maternal care at the age of 3 on children's pro-family values and gender role attitudes. Future research with richer data should focus on which of the channels were the most important and what drives the gender differences that we observe.

Our study is the first to show that family leave policies, typically designed to improve children's well-being, if too generous and replacing preschool education, may negatively impact children's long-term outcomes. The fact that it is primarily the children from low socio-economic backgrounds, with low-educated parents, who are negatively affected suggests that well-intentioned generous family leave policies may actually impede intergenerational mobility and increase educational, social, and economic inequality.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s00148-024-01010-1.

Acknowledgements The authors thank Štěpán Jurajda, Nikolas Mittag, Filip Pertold, participants of SEHO, ESPE, EALE conferences in 2020, 2021, and 2022, editor Terra McKinnish, and three anonymous referees for very useful comments. The responsibility for all remaining errors and conclusions drawn from the data lies entirely with the authors.

Funding Open access publishing supported by the National Technical Library in Prague. This paper was supported by the NPO "Systemic Risk Institute" number LX22NPO5101, funded by European Union - Next Generation EU (Ministry of Education, Youth and Sports, NPO: EXCELES). The earlier stage of this project was financially supported by the Czech Science Foundation, grant number 18-16667S.

Data availability The findings are based on data from the Czech Statistical Office, Labor Force Survey 1994–2020. The original surveys are not publicly available but can be obtained for scientific research purposes only under the conditions specified here https://www.czso.cz/csu/czso/provision_of_microdata_for_scientific_research_purposes_principles_and_conditions for a fee (https://www.czso.cz/csu/czso/price-list-of-services-and-products-zkyy?skupina=25). All codes applied to the original data are available from the authors upon request.

Declarations

Conflict of interest The authors declare no competing interests.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

Bailey MJ, Sun S, Timpe B (2021) Prep school for poor kids: the long- run impacts of head start on human capital and economic self- sufficiency. Am Econ Rev 111:3963–4001

Baker M, Milligan K (2015) Maternity leave and children's cognitive and behavioral development. J Popul Econ 28:373–391

- Bičáková A, Slačálek J, Slavík M (2011) Labor supply after transition: evidence from the Czech Republic. Czech J Econ Finance 61:327–347
- Bičáková A (2016) Gender unemployment gaps in the EU: blame the family. IZA J Eur Labor Stud 5:22
- Bičáková A, Kalíšková K (2019) (Un) intended effects of parental leave policies: evidence from the Czech Republic. Labour Econ 61:101783
- Bičáková A, Kalíšková K (2022) Is longer maternal care always beneficial? The Impact of a Four-Year Paid Parental Leave, IZA Discussion Paper no, p 15640
- Bičáková, A., Kalíšková, K.: Career-breaks and maternal employment in CEE countries. In: Molina, J.A. (ed.), Mothers in the labor market, pp. 159–215 (2022a). Springer International Publishing
- Blanden J, Del Bono E, Hansen K, Rabe B (2021) Quantity and quality of childcare and children's educational outcomes. J Popul Econ 785–828
- Broberg AG, Wessels H, Lamb ME, Hwang CP (1997) Effects of day care on the development of cognitive abilities in 8-year-olds: a longitudinal study. Dev Psychol 33:62–69
- Canaan S (2022) Parental leave, household specialization and children's well-being. Labour Econ 75
- Carneiro P, Loken KV, Salvanes KG (2015) A flying start? Maternity leave benefits and long run outcomes of children. Journal of Political Economy 123:365–412
- Clarke D, Tapia-Schythe K (2021) Implementing the panel event study. Stata J 21:853-884
- Clark-Kauffman E, Duncan GJ, Morris P (2003) How welfare policies affect child and adolescent achievement. Am Econ Rev: Papers and Proceedings 93:299–303
- Cohn DV, Parker K, Livingston G (2014) After decades of decline, a rise in stay-at-home mothers. Pew Research Center
- Cygan-Rehm K, Kühnle D, Riphahn RT (2018) Paid parental leave and families' living arrangements. Labour Econ 53:182–197
- Dahl GB, Lochner L (2012) The impact of family income on child achievement: evidence from the earned income tax credit. Am Econ Rev 102:1927–1956
- Dahl GB, Løken V, Mogstad M, Salvanes KV (2016) What is the case for paid maternity leave? Rev Econ Stat 98:655–670
- Danzer N, Lavy V (2018) Paid parental leave and children's schooling outcomes. Econ J 128:81-117
- Danzer N, Halla M, Schneeweis N, Zweimüller M (2022) Parental leave, (in)formal childcare and long-term child outcomes. J Hum Resour 57:1826–1884
- Dietrichson J, Kristiansen IL, Viinholt BA (2020) Universal preschool programs and long-term child outcomes: a systematic review. J Econ Surv
- Dustmann C, Schönberg U (2012) Expansions in maternity leave coverage and children's long-term outcomes. Am Econ J: Appl Econ 4:190–224
- Ettlerová S, Kuchařová V, Matějková B, Svobodová K, Šťastná A (2006) Postoje a zkušenosti s harmonizací rodiny a zaměstnání rodičů dětí předškolního a mladšího školního vě ku. Výzkumný ústav práce a sociálních věcí, v.v.i
- Felfe C, Lalive R (2018) Does early child care affect children's development? J Public Econ 159(3):33-53
- Felfe C, Nollenberger N, Rodríguez-Planas N (2015) Can't buy mommy's love? Universal childcare and children's long-term cognitive development. J Popul Econ 28:393–422
- Fodor E (2005) Women at work: the status of women in the labour markets of the Czech Republic, Hungary and Poland. JUNRISD Occasional paper 3/2005
- García JL, Heckman JJ, Leaf DE, Prados MJ (2020) Quantifying the life-cycle benefits of an influential early-childhood program. J Pol Econ 128:2502–2541
- Gertler PJ, Martinez S, Premand P, Rawlings LB, Vermeersch CMJ (2016) Impact evaluation in practice, 2nd edn. Inter-American Development Bank and World Bank, Washington, DC
- Ginja R, Karimi A, Xiao P (2023) Employer responses to family leave programs. Am Econ J: Appl Econ 15(1):107–135
- Havnes T, Mogstad M (2011) No child left behind: subsidized child care and children's long-run outcomes. Am Econ J: Econ Pol 3(5):97–129
- Havnes T, Mogstad M (2015) Is universal child care leveling the playing field? J Public Econ 127:100-114
- Hoxby CM (2021) Advanced cognitive skill deserts in the United States: their likely causes and implications. Brookings Papers on Economic Activity 2021 (mar). 317–351
- Huebener M (2016) Parental leave policies and child development: a review of empirical findings. DIW Berlin
- Huebener M, Kuehnle D, Spiess CK (2019) Parental leave policies and socio-economic gaps in child development: evidence from a substantial benefit reform using administrative data. Labour Econ 61:101754

- Kantorova V (2004) Education and entry into motherhood: the Czech Republic during state socialism and the transition period. Demographic Research Special Collections 3:245–274
- Kawarazaki Hikaru (2023) Early childhood education and care: effects after half a century and their mechanisms. J Popul Econ 36:2725–2797
- Kleven H, Landais C, Søgaard JE (2019) Children and gender inequality: evidence from Denmark. Am Econ J: Appl Econ 11:181–209
- Kuchařová V, Bareš P, Hohne S, Nešporová O, Svobodová K, Šťastná A, Plasová B, Žáčková L (2009) Péče o děti předškolního a raného školního věku. Výzkumný ústav práce a sociálních věcí, v.v.i
- Liu Q, Skans ON (2010) The duration of paid parental leave and children's scholastic performance. B.E. J Econ Anal Pol 10(1)
- Loeb S, Bridges M, Bassok D, Fuller B, Rumberger R (2007) How much is too much? The influence of preschool centers on children's social and cognitive development. Econ Educ Rev 26:52–66
- Nicoletti C, Salvanes KG, Tominey E (2023) Mothers working during preschool years and child skills: does income compensate? J Labor Econ 41(4):389–429
- Pertold-Gebicka B (2020) Parental leave length and mothers' careers: what can be inferred from occupational allocation? Appl Econ 52:879–904
- Pronzato CD (2009) Return to work after childbirth: does parental leave matter in Europe? Rev Econ Household 7(5):341–360
- Rasmussen AW (2010) Increasing the length of parents' birth-related leave: the effect on children's longterm educational outcomes. Labour Econ 17(1):91–100
- Rossin-Slater M (2018) In: Susan L, Argys ALM, Hoffman SD (eds) Maternity and family leave policy. Oxford University Press. p 323–342
- Rude B (2023) Middle-run educational impacts of comprehensive early childhood interventions: evidence from a pioneer program in Chile. J Popul Econ, vol. forthcoming
- Saxonberg S, Sirovátka T (2006) Failing family policy in post-communist Central Europe. J Comp Policy Anal: Res Pract 8:185–202
- Schönberg U, Ludsteck J (2014) Expansions in maternity leave coverage and mothers' labor market outcomes after childbirth. J Labor Econ 32:469–505
- Sobotka T (2003) Re-emerging diversity: rapid fertility changes in Central and Eastern Europe after the collapse of the communist regimes. Population 58:451
- Steinhauer A (2018) Working moms, childlessness, and female identity. LIEPP Working Paper, vol 79

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.